FORM PTO-1594 (Rev. 6-93)		REC(06-22	2-2000	HEET	U.S. DEPARTME	NT OF COMMERC
OMB No. 0651-0011 (exp. 4/94)	RE			1 1800 1868	1		nd Trademark Offi
Tab settings ⇔⇔⇔ ▼	100	▼			↑ (0,	1600 +	•
To the Honorable (Commissi	oner of Pate	1013	87586	ched original o	documents or cop	y thereof.
1. Name of conveying	party(ies	s):		2. Name and	address of rec	eiving party(ies)	
Monon Corporat	ion			Name:	HPA Mo	onon, Inc.	
				Internal Add	dress: One Wa	ater Tower D	rive
☐ Individual(s) ☐ General Partnership		☐ Associatio☐ Limited Pa			ress: P.O. H		
Corporation-StateOther						State: <u>IN</u>	
Additional name(s) of conve	ing party(ies) attached?	Yes □ No			p	
3. Nature of conveyand	ce:			☐ General	Partnership		
☐ Assignment		□м	erger		tion-State		
☐ Security Agreem		Q C	nange of Name	•			
Other Asset	Purcha	ise Agreemer	ıt	If assignee is not do is attached:	miciled in the United	d States, a domestic re	presetative designat
Execution Date: Febr	ruary 2	20, 1997		(Designations must Additional name(s) &		ment from assignment) ned? 🖸 Yes 💢 No	,
4. Application number(s) or pa	tent number(s):					
A. Trademark Appl	ication N	lo.(s)		ł		on No.(s)	
		A	dditional numbers at	tached? ☐ Yes ☐ N			
5. Name and address concerning docume			pondence	3	per of applications involved:	ons and	3
Name: Miles A.	Zvi,	Esq.					
Internal Address:						\$ <u>90.00</u>	
				X Previ	lously paid		
Fairweather 8	Geral	dson		☐ Authori	zed to be char	rged to deposit a	account
Street Address: _55	East	Monroe - Su	ite 4200	8. Deposit ac	count number:	:	
City: Chicago	s	tate: <u>IL</u>	ZIP: 60603				
	·····		DO NOT US	Attach duplic SE THIS SPACE	cate copy of this p	page if paying by de	posit account)
···							
Statement and sign: To the best of my ki the original docume	nowledg	e and belief, the	e foregoing inform	nation is true and	correct and an	y attached copy	ı is a true copy
Miles A. Zvi			Mi	2 6 3.		6/15	100

Name of Person Signing

Signature

Total number of pages including cover sheet, attachments, and document:

Date

REEL: 002091 FRAME: 0656

Un3/54/00

FORM PTO-1618A Expires 06/30/99



U.S. Department of Commerce Patent and Trademark Office TRADEMARK

	14 Million 364
	101303640
RECORDATIO	
	MARKS ONLY
	: Please record the attached original document(s) or copy(ies).
Submission Type	Conveyance Type
X New	Assignment License
Resubmission (Non-Recordation) Document ID #	Security Agreement Nunc Pro Tunc Assignme
Correction of PTO Error Reel # Frame #	Merger Month Day Year 02-20-97
Corrective Document	Change of Name
Reel # Frame #	X Other ASSET PURCHASE AGREEMENT
Conveying Party	Mark if additional names of conveying parties attached Execution Date
Name MONON CORPORATION	Month Day Yea
Name Honon Conformitor	
Formerly	
Individual Consed Body within	Limited Body and Table 1
Individual General Partnership	Limited Partnership X Corporation Associatio
Other	
Citizenship/State of Incorporation/Organiza	dian.
Receiving Party	
Receiving Party	Mark if additional names of receiving parties attached
Name HPA MONON, INC.	
DDA/AKA TA	
DBA/AKA/TA	
Composed of	
Address (line 1) One Water Tower Drive	
Address (line 1) Une Water Tower Drive	
Address (line 2) Post Office Box 635	
Address (line 3) Monon	Indiana 47959
City	Indiana 47959 State/Country Zip Code
Individual General Partnership	Limited Partnership If document to be recorded is an
Individual General Partnership	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an
Individual General Partnership X Corporation Association	Limited Partnership If document to be recorded is an assignment and the receiving party is
Individual General Partnership	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an appointment of a domestic representative should be attached. (Designation must be a separate
Individual General Partnership X Corporation Association Other	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an appointment of a domestic representative should be attached. (Designation must be a separate document from Assignment.)
Individual General Partnership X Corporation Association Other Citizenship/State of Incorporation/Organizate	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an appointment of a domestic representative should be attached. (Designation must be a separate document from Assignment.)
Individual General Partnership X Corporation Association Other Citizenship/State of Incorporation/Organizate	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an appointment of a domestic representative should be attached. (Designation must be a separate document from Assignment.)
Individual General Partnership X Corporation Association Other Citizenship/State of Incorporation/Organizat	Limited Partnership If document to be recorded is an assignment and the receiving party is not domiciled in the United States, an appointment of a domestic representative should be attached. (Designation must be a separate document from Assignment.)

Mail documents to be recorded with required cover sheet(s) information to:
Commissioner of Patents and Trademarks, Box Assignments , Washington, D.C. 20231

TRADEMARK

REEL: 002091 FRAME: 0657

FORM PTO- Expires 06/30/99 OMB 0651-0027	1618B Page 2	U.S. Department of Commerce Patent and Trademark Office TRADEMARK				
	epresentative Name and Address Enter for the first Receiving Pa	rty only.				
Name [
Address (line 1)						
Address (line 2)						
Address (line 3)						
Address (line 4)						
Correspond	lent Name and Address Area Code and Telephone Number 312-269-88	95				
Name	Miles A. Zvi					
Address (line 1)	SEYFARTH, SHAW, FAIRWEATHER & GERALDSON					
Address (line 2)	55 East Monroe Street, Suite 4200					
Address (line 3)	Chicago, IL 60603					
Address (line 4)						
Pages Enter the total number of pages of the attached conveyance document # 97						
	Application Number(s) or Registration Number(s) Mark if act of Trademark Application Number or the Registration Number (DO NOT ENTER BOTH numbers for the Registration Number (DO NOT ENTER BOTH numbers for the North Number (DO NOT ENTER BOTH numbers (DO NOT ENTER BOTH n	Iditional numbers attached or the same property).				
Trac	lemark Application Number(s) Registration Num	`				
	726,679 1,584,53	6 1,864,178				
Number of F	Properties Enter the total number of properties involved. # 3					
Fee Amoun	t Fee Amount for Properties Listed (37 CFR 3.41): \$ 96	0.00				
Method of Deposit A	f Payment: Enclosed X Deposit Account					
(Enter for p	ayment by deposit account or if additional fees can be charged to the account.) Deposit Account Number: #					
	Authorization to charge additional fees: Yes	No No				

To the best of my knowledge and belief, the foregoing information is true and correct and any attached copy is a true copy of the original document. Charges to deposit account are authorized, as

Statement and Signature

Miles A. Zvi

indicated herein.

Name of Person Signing

TRADEMARK

REEL: 002091 FRAME: 0658

PURCHASE AGREEMENT

among

HPA ASSET, LLC

and

MONON CORPORATION and MONON TRAILER, INC., Debtors and Debtors in Possession,

February 20, 1997

Doc#:DS3:18979.10 25-090

EXHIBIT 1

TABLE OF CONTENTS

RECITALS		•	•	•	1
ARTICLE 1	DEFINITIONS	•	•		1
ARTICLE 2	TRANSFER OF ASSETS AND LIABILITIES	•	•	•	8
2.1	Sale of Assets				8
2.2	Excluded Assets	•		•	9
2.3	Assumed Liabilities	•		. 1	0
2.4	Excluded Assets	•	•	. 1	. 1
ARTICLE 3	PURCHASE PRICE	•	•	. 1	. 2
3.1	Closing Date Consideration	•	•	. 1	. 2
3.2	Allocation of Purchase Price	•	٠	1	. 2
3.3	Payment of Transfer Taxes			. 1	2
3.4	Payment of Transfer Taxes	•	•	. 1	. 3
ARTICLE 4	CLOSING	•	•	. 1	. 3
4.1	Time and Place	•		. 1	. 3
4.2	Closing Deliveries of Seller			. 1	. 3
	Closing Deliveries of Buyer				
ARTICLE 5	REPRESENTATIONS AND WARRANTIES OF SELLER	•	•	. 1	. 4
5.1	Organization of Seller	• ,		. 1	4
5.2	Authority to Execute and Perform Agreement		-	. 1	4
5.3	Consents and Approvals; No Violations	. ,	•	. 1	4
5.4	Permits			. 1	5
5.5	Litigation			. 1	5
5.6	Contracts	• (. 1	5
5.7	No Broker			. 1	5
5.8	Affiliate Assets			. 1	5
5.9	Title to Assets			. 1	5
5 10	Other Assets	•	•	· -	5
5 11	Benefit Plans	•			5
5.12	No Known Breach or Impediment	•		. 1	6
ARTICLE 6	REPRESENTATIONS AND WARRANTIES OF BUYER .	· •	, ,	. 1	6
6.1	Organization of Buyer			. 1	6
6.2	Authority to Execute and Perform Agreement	_			
6.3	No Broker		, .	. 1	6
6.4	No Broker			. 1	6
6.5	No Known Breach or Impediment		•	. 1	7
አውጥተሮፒፍ 7	CONDUCT PRIOR TO CLOSING			1	7

i

Doc#:DS3:18979.10 25-090

8.1 Change and Use of Seller's Name					
7.2 Litigation	7.1	Conduct of Business	•	•	. 17
7.3 Access to Information and Employees	7.2	Litigation			. 17
7.4 Insurance	7.3	Access to Information and Employees		_	. 18
### AGREEMENT OF THE PARTIES AS TO CERTAIN PRE-CLOSING ACTIONS	7.4	Insurance	_	_	. 18
ARTICLE 8 AGREEMENT OF THE PARTIES AS TO CERTAIN PRE-CLOSING ACTIONS	7 5	Bankruntcy Filings	•	•	18
### REF-CLOSING ACTIONS	7.5	banki upecy riffings	•	•	. 10
### REF-CLOSING ACTIONS	ADMIGLE O	ACDEEVENM OF MUE DADMIES AS MO SEDMAIN			
8.1 Change and Use of Seller's Name	ARTICLE 6				4.0
8.2 Filings		PRE-CLOSING ACTIONS	•	•	. 19
8.2 Filings		Change and the of College Many			
### RECORDITIONS PRECEDENT TO THE OBLIGATION OF BUYER TO CLOSE		Change and use of Seffer's Name	•	•	. 19
### RECORDITIONS PRECEDENT TO THE OBLIGATION OF BUYER TO CLOSE	8.2	filings	•	•	. 19
ARTICLE 9 CONDITIONS PRECEDENT TO THE OBLIGATION OF BUYER TO CLOSE	8.3	Access to Leased Property	•	•	19
### BUYER TO CLOSE	8.4	Conditions	•	•	20
### BUYER TO CLOSE			_		
9.1 Representations, Warranties and Covenants	ARTICLE 9				
9.2 No Orders		BUYER TO CLOSE	•	•	. 20
9.2 No Orders					
9.4 Title to the Assets		Representations, Warranties and Covenants	•	•	. 20
9.4 Title to the Assets	9.2	No Orders	•	•	. 20
9.4 Title to the Assets	9.3	Approvals of Congress Financial			. 20
9.5 Bankruptcy Court Approval	9.4	Title to the Assets			. 21
9.6 Required Filings/Notices				_	. 21
9.7 Financing					
9.8 Rosby Corporation Assets	9.7	Financing	•	•	21
9.9 Minimum Inventory Levels	0.7	Pochy Corneration Assets	•	•	. 21
ARTICLE 10 CONDITIONS PRECEDENT TO THE OBLIGATION OF SELLERS TO CLOSE					
SELLERS TO CLOSE	9.9	minimum inventory Levels	•	•	. 22
SELLERS TO CLOSE		A CONDITIONS PRESERVE TO MUE ARTICIMION C	.		
10.1 Representations, Warranties and Covenants	ARTICLE IC	U CONDITIONS PRECEDENT TO THE OBLIGATION O	ľ		•
10.2 No Orders		SELLERS TO CLOSE	•	•	. 22
10.2 No Orders					
10.4 Bankruptcy Court Approval	10.1	Representations, Warranties and Covenants	•	•	. 22
10.4 Bankruptcy Court Approval	10.2	No Orders	•	•	. 22
10.4 Bankruptcy Court Approval	10.3	Approvals	•		. 22
10.5 HSR Filings	10.4	Bankruptcy Court Approval	•	•	. 23
10.6 Purchase Price	10.5	HSR Filings	•		. 23
10.9 Creditors Committee Approval	10.6	Purchase Price		•	. 23
10.9 Creditors Committee Approval	10.7	Assumption of Liabilities			. 23
10.9 Creditors Committee Approval	10.8	Commitment from Congress Financial		_	. 23
ARTICLE 11 BANKRUPTCY COURT APPROVALS; OVERBIDDING PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT	10.9	Creditors Committee Approval		•	23
ARTICLE 11 BANKRUPTCY COURT APPROVALS; OVERBIDDING PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT	10.10	O Completion of Auction	•	•	23
ARTICLE 11 BANKRUPTCY COURT APPROVALS; OVERBIDDING PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT	10.10	1 Pochu Corporation Assets	•	• •	, 23
PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT	10.11	I ROSDY Corporation Assets	•	•	. 23
PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT	DOTATE 11	1 PANTEDITORCY COLLEGE ADDROVATE OVERDETORING			
REIMBURSEMENT	ARTICLE II				
11.1 Sale Order					
11.4 Overbidding Procedures 24		REIMBURSEMENT	•	• •	. 23
11.4 Overbidding Procedures 24					~ ~
11.4 Overbidding Procedures 24	11.1	Sale Order	•	• •	23
11.4 Overbidding Procedures 24	11.2	Failure to Obtain Approval	•	• •	. 24
11.4 Overbidding Procedures 24	11.3	Competing Bid	•		. 24
11.5 Payment of Break-Up Expenses and Fee 24	11.4	Overbidding Procedures	•		. 24
	11.5	Payment of Break-Up Expenses and Fee	•		24

ii

Dee#:DS3:18979.10 25-090

ARTICLE 12	[ERMINATION	N OF AGREEMENT	25
12.1 Ter	cmination .		25
12.2 Rer	nedies in t	the Event of Failure to Close	26
		rmination	
ARTICLE 13 (CONDITION C	OF ASSETS	27
ARTICLE 14 1	ISCELLANEC	ous	27
14.1 Ce	ertain Tax	Matters	27
14.2 En	ployees .		28
14.3 Ex	menses		28
14.4 Cc	ollections		28
14.5 No	tice to Th	hird Parties	29
14.6 Do	cuments ar	nd Other Information	29
14.7 Di	spute Resc	olution	29
14.8 Se	verability	y	30
14.0 Eu	rther Aces	urances	30
		ement	
14.11 EN	icire Agree	Prondroute: Non-Contractual	32
14.12 Ma	modice	Amendments; Non-Contractual	22
14 12 Co	medles		22
14.13 GO	verning La	aw; Jurisdiction of Court	32
14.14 00	uncerparts	s	32
14.15 81	naing Eile	ect; No Assignment	32
		Beneficiary	
14.17 Va	riations i	in Pronouns	33
		d Schedules	
14.19 He	adings and	d Table of Contents	33
Exhibit	A	Bill of Sale	
Exhibit	B	Permitted Liens	
Exhibit	C	Sale Order	
Exhibit	D	Permitted Liens Sale Order Overbidding Procedures	
Schedule	2.1	Designated Contracts	
Schedule	2.2	Other Excluded Assets	
Schedule	5.2	Exceptions to Sellers' Authority	
		to Execute and Perform Agreemen	t
Schedule	5.3	Other Required Consents and	
		Approvals for Sellers	
Schedule	5.4	Permits	
Schedule		Litigation	
Schedule		Affiliate Assets	
Schedule		Other Material Assets	
Schedule		Employee Benefit Plans	
Schedule		Other Required Consents and	
Schedule	0.4	Approvals for Buyer	
		White ages for paler	

PURCHASE AGREEMENT

This PURCHASE AGREEMENT (the "Agreement") is entered into as of the 20th day of February, 1997 by and among HPA ASSET, LLC, a Delaware limited liability company ("Buyer"), and MONON CORPORATION, a Delaware corporation ("Moron"), and MONON TRAILER, INC., a Delaware corporation ("MTI"), as debtors and debtors in possession (each a "Seller" and, collectively, "Sellers").

RECITALS

WHEREAS, on September 25, 1996 ("Petition Date"), certain creditors of Monon commenced Case No. 96-40623 by filing an involuntary petition under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101 et seq. (the "Bankruptcy Code"), in the United States Bankruptcy Court for the Northern District of Indiana (the "Bankruptcy Court"), and on October 11, 1996, Monon consented to chapter 11 relief, and on the same date, MTI commenced Case No. 96-12098 by filing a voluntary petition under chapter 11 of the Bankruptcy Code, which case is being jointly administered with Case No. 96-40623 (collectively, the "Chapter 11 Case");

WHEREAS, Sellers have ceased operation of their ongoing Businesses (as hereinafter defined);

WHEREAS, Sellers desire to sell and assign to Buyer, and Buyer desires to purchase from Sellers, all of Sellers' right, title and interest in and to the Assets (as defined below) upon the terms and subject to the conditions of this Agreement; and

WHEREAS, Sellers understand that Buyer is not a successor to the Business of either Seller.

NOW THEREFORE, in consideration of the mutual promises and agreements set forth herein, and intending to be legally bound, Buyer and Sellers agree as follows:

ARTICLE 1

DEFINITIONS

As used throughout this Agreement, the following terms have the following meanings:

"Accounts Receivable" shall mean any right of Sellers to payment for goods sold or services rendered.

"Affiliate" shall mean, with respect to any Person, any other Person controlling, controlled by or under common control with such Person. For purposes of this definition, "control" shall mean the power to direct, or

Doc#: DS3:18979.10 25-090

cause the direction of, the management or policies of any Person, whether through ownership of securities, by contract or otherwise.

"Affiliate Assets" shall have the meaning set forth in Section 9.8.

"Agreement" shall have the meaning set forth in the Preamble to this Agreement.

"Allocation Schedule" shall have the meaning set forth in Section 3.2.

"Approvals" shall mean all consents, authorizations, qualifications or other approvals.

"Assets" shall have the meaning set forth in Section 2.1.

"Assumed Liabilities" shall mean the Liabilities assumed by Buyer pursuant to Section 2.3.

"Bankruptcy Claims" shall have the meaning set forth in Section 2.2(e).

"Bankruptcy Code" shall have the meaning set forth in the Recitals to this Agreement.

"Bankruptcy Court" shall have the meaning set forth in the Recitals to this Agreement.

"Bankruptcy Schedules" shall mean the lists, statements and schedules heretofore filed by Sellers in the Chapter 11 Case pursuant to Rule 1007 of the Federal Rules of Bankruptcy Procedure.

"Benefit Plan" shall mean any employee benefit plan, agreement, arrangement, policy or commitment of any type which is an employment, consulting or other compensation agreement or funding arrangement, including, without limitation, any plan described in Section 3(3) of ERISA, whether or not covered by ERISA and any other employment or consulting agreement, pension, profit sharing, bonus, savings, retirement, stock purchase, stock option, insurance, health care, severance and other employee fringe benefit, incentive compensation or deferred salary plan, under which either Seller has, or in the future could have, directly, or indirectly through an ERISA Affiliate, any actual or contingent liability or obligation.

"Bill of Sale" shall mean a Bill of Sale and Assignment and Instrument of Assumption executed by Sellers

Doc#:DS3:18979.10 25-090

and Buyer, substantially in the form attached hereto as Exhibit A.

"Books and Records" shall mean the books and records owned by Sellers (or true and complete copies thereof), including all computerized print-outs of books and records owned by Sellers, including, without limitation, all such books and records relating to the purchase or sale of the Assets in the ordinary course of business, supplies and services for the Business or dealings with customers of the Business; provided, however, that "Books and Records" shall not include any books, records, documentation or information (a) relating to Excluded Assets or prepared or developed in connection with the Contemplated Transactions, (b) the disclosure of which to Buyer or its Representatives would result in a breach of any confidentiality or nondisclosure agreement between Sellers or any of Sellers' Affiliates, on the one hand, and any one or more third parties (other than Sellers and their Affiliates), on the other hand, provided that, Sellers shall use reasonable efforts to obtain the consent of such third party to disclose such information to Buyer or its Representatives, or (c) that are property of any Person other than Sellers.

"Break-Up Expenses" shall have the meaning set forth in Section 11.5.

"Break-Up Fee" shall have the meaning set forth in Section 11.5.

"Business" shall mean, with respect to Sellers, the business currently conducted by each Seller.

"Business Day" shall mean any day excluding Saturday, Sunday and any day which is a legal holiday under the laws of the State of Indiana or is a day on which banking institutions located in such state are authorized or required by law or other government action to close.

"Buyer" shall have the meaning set forth in the Preamble to this Agreement.

"Chapter 11 Case" shall have the meaning set forth in the Recitals to this Agreement.

"Claims" shall mean all actions, suits, claims, demands, charges or complaints or legal, administrative or arbitral proceedings or investigations, whether pending, threatened, known or unknown, inchoate, contingent, liquidated or unliquidated.

"Closing" shall have the meaning set forth in Section 4.1.

Doc#:DS3:18979.10 25-090

"Closing Date" shall have the meaning set forth in Section 4.1.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Collective Bargaining Agreement" shall mean any contract or agreement including but not limited to any memorandum of understanding, letter, side letter or stipulation, whether oral or written, between either Seller or any of Sellers' Affiliates, or any Person on behalf of the foregoing, and any union or labor organization representing any Employees.

"Competing Bid" shall have the meaning set forth in Section 11.3.

"Competing Bidder" shall have the meaning set forth in Section 11.3.

"Congress Debt" shall mean the indebtedness owing by Sellers to Congress Financial Corporation (Central) ("Congress Financial") in an amount equal to the lesser of (a) the entire outstanding balance of such indebtedness or (b) \$12,000,000.

"Contemplated Transactions" shall mean the transactions contemplated by this Agreement.

"Contracts" shall mean all contracts, agreements, indentures, notes, bonds, loans, instruments, leases, sub-leases, deeds of trust, conditional sales contracts, mortgages, franchises, licenses, commitments or other binding arrangements, express or implied, currently in effect to which either Seller is a party or a third party beneficiary or under which either Seller has rights or by which either Seller or the Assets is or are bound or under which either Seller otherwise has obligations.

"Deposit" shall mean the sum of \$700,000 paid by Buyer to Sellers upon the execution of this Agreement plus interest earned thereon from the date such amount was paid to Sellers until the earlier of (a) the Closing Date or (b) the date upon which Sellers either (i) are permitted to retain the Deposit pursuant to Section 12.2(a) or (ii) pay over the Deposit to Buyer.

"Designated Contracts" shall mean the Contracts listed on Schedule 2.1 that Buyer determines to assume, any or all of which Sellers agree to assume and assign to Buyer or its designee pursuant to Section 365 of the Bankruptcy Code provided that Buyer gives Sellers written notice of

Doc#:DS3:18979.10 25-090

Buyer's desire to have Sellers do so by no later than March 11, 1997.

"DOJ" shall have the meaning set forth in Section 8.2.

"Employee" shall mean any current employee of either Seller immediately prior to the Closing.

"Employee Benefit Plan" shall mean any Benefit Plan which covers any current or former officer, director, consultant or employee (or beneficiary thereof) of either Seller, or any Benefit Plan which is covered by ERISA and in respect of which either Seller is (or if such plan were terminated, would, at such time, under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"Employment Liability" shall mean any liability or obligation of any kind or nature to the extent relating to, arising out of, or otherwise with respect to (i) the employment or termination of employment by either Seller or an ERISA Affiliate of any current or former officer, director, consultant or employee of either Seller or any ERISA Affiliate, or (ii) any Benefit Plan (including any Employee Benefit Plan), including, without limitation, any plan contribution to, any excise tax arising with respect to, or any withdrawal from or termination of, any Benefit Plan.

"Environment" shall mean any indoor or outdoor ambient air, surface water, ground water, drinking water, building surface, material surface, land surface or subsurface strata.

"Environmental Liability" shall mean any liability or obligation arising out of, in connection with or pursuant to any violation of, or Claim made, in connection with or related to, a Law for the protection of the Environment, including, without limitation, liability for investigatory costs, oversight costs, remediation and cleanup costs, governmental or private response costs and cost recovery actions, natural resource damages, property damages, personal injuries, consequential economic damages, administrative, civil or criminal penalties or forfeitures, and attorneys' fees or other costs of defending a Claim asserting liability under any Law for the protection of the Environment.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

Doc#:DS3:18979.10 25-090

"ERISA Affiliate" shall mean any entity, whether or not incorporated, which is, or in the past five years was, under common control with Seller within the meaning of Section 4001 of ERISA or is part of a group which includes either Seller and which is treated as a single employer under Section 414 of the Code;

"Excluded Assets" shall have the meaning set forth in Section 2.2.

"Excluded Liabilities" shall have the meaning set forth in Section 2.4.

"FTC" shall have the meaning set forth in Section 8.2.

"Governmental Body" shall mean any court, governmental or regulatory agency, authority, body or arbitrator of any state, local, federal or foreign government.

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations of the FTC promulgated thereunder.

"Intellectual Property" shall have the meaning set forth in Section 2.1(b).

"Law" shall mean any federal, state, local or foreign law, statute, order, ordinance or regulation.

"Liabilities" shall mean any and all debts, liabilities and obligations, whether accrued, contingent, disputed, undisputed, secured, unsecured, liquidated, unliquidated, matured or unmatured, including, without limitation, (a) those arising under any Law, Order or Contract; (b) all Environmental Liabilities; and (c) all Employment Liabilities.

"Lien" shall mean any lien, security interest, mortgage, pledge, right of first or last refusal or offer, lease, hypothecation, assignment, deposit arrangement, charge, conditional sale, encumbrance or any other security agreement or preferential arrangement of any kind whatsoever.

"Material Adverse Effect" shall mean the incurrence by Buyer in the conduct of the Business after the Closing of (a) out-of-pocket costs and expenses of \$100,000, in the aggregate, or (b) depreciation in value of the Assets of \$250,000, in the aggregate.

Doc#:DS3:18979.10 25-090

"Monon" shall have the meaning set forth in the Recitals to this Agreement.

"MTI" shall have the meaning set forth in the Recitals to this Agreement.

"Order" shall mean any order, judgment, injunction, award, decree or writ of a Governmental Body.

"Permits" shall mean all permits, licenses, certificates, franchises and other authorizations, consents and approvals of any Governmental Body necessary for any Person to conduct its Business.

"<u>Permitted Liens</u>" shall mean any Liens: (a) to be discharged or released on or before the Closing Date; or (b) set forth on Exhibit <u>B</u> hereto.

"Person" shall mean any corporation, partnership, firm, joint venture, individual, association, trust, unincorporated organization or other entity.

"Petition Date" shall have the meaning set forth in the Recitals to this Agreement.

"Property Taxes" shall mean the real estate taxes, if any, (a) owing as of the Closing Date to the Treasurer, White County, Indiana, which constitute a lien on the Assets, in an amount up to, but no greater than, \$170,000, and (b) such real estate taxes for 1996 and subsequent years with respect to the Assets.

"<u>Purchase Price</u>" shall mean \$2,500,000 less the Deposit.

"Representatives" means, with respect to any party, the managers, members, directors, officers and employees of such party or its subsidiaries and its accountants, counsel, financial advisors and other agents.

"Sale Order" shall have the meaning set forth in Section 11.1.

"Seller" or "Sellers" shall have the meaning set forth in the Preamble to this Agreement.

"Taxes" shall mean all federal, state, county, local, foreign and other taxes for which either Seller is liable (including, without limitation, income, profits, premium, estimated, excise, sales, use, occupancy, gross receipts, franchise, ad valorem, severance, capital levy, production, transfer, withholding, employment, unemployment compensation, payroll related and property taxes, import

Doc#:DS3:18979.10 25-090

duties and other governmental charges and assessments), whether or not measured in whole or in part by net income, and including deficiencies, interest, additions to tax or interest, and penalties with respect thereto, and including expenses associated with contesting any proposed adjustment related to any of the foregoing and any liabilities, obligations and expenses pursuant to any tax sharing, tax indemnification, tax allocation or similar arrangements to which either Seller is a party.

"Transfer Taxes" shall mean all stamp, transfer, documentary, sales, use, registration, recordation and other such taxes and fees incurred in connection with this Agreement and the Contemplated Transactions, but shall exclude other Taxes.

"Uniform Commercial Code" shall mean the version of the Uniform Commercial Code in effect in the State of Indiana or otherwise applicable to the relevant activity or event.

ARTICLE 2

TRANSFER OF ASSETS AND LIABILITIES

- 2.1 <u>Sale of Assets</u>. Subject to the terms and conditions set forth in this Agreement, and pursuant to Sections 105, 363 and 365 of the Bankruptcy Code, at the Closing, Sellers shall sell, assign, transfer and deliver to Buyer, free and clear of any and all Liens and Claims, except Permitted Liens and except as described in Section 2.3, and Buyer shall purchase, acquire and take assignment and delivery of all right, title and interest of Sellers, in and to the following assets and properties owned by Sellers (collectively, but excluding the Excluded Assets, the "<u>Assets</u>"):
- (a) those assets and properties reflected on the books and records of Sellers, together with the following assets and properties, whether or not so reflected: (i) all real property, plant and fixtures, (ii) all machinery, equipment, furniture and other items of tangible personal property (including construction in progress) owned by Sellers, (iii) all inventories, wherever located, (iv) all trucks, automobiles and other vehicles (including any and all rights to redeem any of the foregoing Assets from secured parties pursuant to the applicable Uniform Commercial Code or otherwise), (v) all Accounts Receivable and other amounts owing to Sellers, (vi) goodwill and (vii) general intangibles;

Doc#:DS3:18979.10 25-090

- (b) all names, trade names (including, but not limited to, the name "Monon"), trademarks, service marks, labels, copyrights, patents (including applications for, rights to acquire and other rights with respect to any of the foregoing), trade secrets, know-how and all other intellectual property held or used by Sellers, and all rights related thereto under the trademark and related laws of every country and jurisdiction throughout the world, now or hereafter known, including, without limitation, all customer lists, product designs, style data, patterns, prototypes, technical data, production outlines, specifications, blueprints, drawings, sales history, bills of materials, mailing lists, telemarketing programs and demographies (collectively, together with subsections (h) and (i) below, the "Intellectual Property");
- (c) the Permits (to the extent transferable) and Designated Contracts;
 - (d) all Books and Records;
- (e) all Claims of Sellers as of the Closing Date, including, without limitation, any Claims under express or implied warranties from suppliers of Sellers with respect to the Assets, but excluding the Bankruptcy Claims;
- (f) all rights under any letter of credit issued in favor of the Seller (to the extent transferable);
- (g) all sales, advertising and promotional
 (i) audiovisual material, (ii) video masters and variations,
 (iii) literature, photography and color separations;
- (h) all molds, tools, dies, assembly equipment and associated methodology necessary to manufacture any products, including, without limitation, testing, inspection and quality control processes and techniques, material specifications, technology, processes, other manufacturing and technical information and other archived materials;
- (i) to the extent transferable, existing telephone numbers, telecopier numbers and telex numbers used by Sellers and relating to the Assets; and
- (j) all media ad data bases and history and projection reports used in the Business.
- 2.2 <u>Excluded Assets</u>. Anything in this Agreement to the concrary notwithstanding, there shall be excluded from the Assets, for all purposes, the following assets of Sellers (collectively, the "<u>Excluded Assets</u>"):
 - (a) all cash of Sellers;

Doc#:D\$3:18979.10 25-090

- (b) any retainers held by any attorney or other adviser to Sellers;
- (c) Tax records, Tax Claims and rights to Tax refunds, other than with respect to Transfer Taxes;
 - (d) Contracts not assumed by Buyer;
- (e) all assets of any Benefit Plan, including any Employee Benefit Plan;
- (f) minute books and other records relating to either Seller's incorporation, and Sellers' stock ledgers and stock transfer books;
- (g) Claims of Sellers, as debtors or debtors in possession in the Chapter 11 Case, to recover preferences or fraudulent conveyances or in other avoidance actions or proceedings pursuant to Sections 544, 545 or 547-553 of the Bankruptcy Code (the "Bankruptcy Claims");
- (h) rights of Sellers under this Agreement and all other agreements, instruments and documents executed in connection herewith;
 - (i) the stock of MTI;
 - (j) real estate owned by MTI of Detroit, Inc.;
- (k) those assets and rights set forth on Schedule 2.2; and
- (1) any assets, properties or rights that are not owned by Sellers, whether or not used by Sellers or owned by an Affiliate of Sellers; provided, however, that (i) any "Patents owned by Rosby Corporation" and (ii) U.S. Patent No. 4,904,017 listed on Schedule 2.2 which are hereafter determined, by final court order or otherwise, to be the property of either Seller, shall be licensed by Sellers (x) to Buyer or, at Buyer's direction, to any other Person who is a permitted licensee under, and (y) on terms and conditions substantially similar to those contained in, any license agreement entered into prior to the Closing between Rosby Corporation and Buyer pursuant to Section 9.8 of this Agreement.
- 2.3 <u>Assumed Liabilities</u>. Subject to the terms and conditions set forth in this Agreement, and pursuant to Sections 105, 363 and 365 of the Bankruptcy Code, at the Closing, Buyer shall assume and agree to perform in accordance with their respective terms, (i) the Designated Contracts, including payment of amounts required to cure any defaults necessary for the assumption and assignment thereof

more of the

Doc#:DS3:18979.10 25-090

pursuant to Section 365(b)(1) of the Bankruptcy Code, whether such amounts accrued before or after the Petition Date, (ii) the Congress Debt and (iii) the Property Taxes (collectively, the "Assumed Liabilities").

- 2.4 Excluded Liabilities. Except as otherwise provided in Section 2.3, Buyer expressly does not assume, and Sellers agree that, without affecting the applicability of the Bankruptcy Code to their rights and obligation with respect to third parties, Sellers shall be liable for and shall retain (collectively, the "Excluded Liabilities"):
- (a) all Liabilities, obligations and expenses relating to the Excluded Assets;
- (b) all Liabilities, obligations and expenses of any kind or nature relating to Taxes required to be paid by or with respect to Sellers or any consolidated, combined or similar group of which either Seller is or has been a member and, with respect to the Assets, for any period ending on or before the Closing Date and any Taxes payable by sellers in connection with the Contemplated Transactions other than as set forth in Section 3.3;
 - (c) all Employment Liabilities;
- (d) all Liabilities, obligations and expenses of any kind or nature with respect to, or arising out of, any Collective Bargaining Agreement;
- (e) all Liabilities, obligations and expenses of any kind or nature with respect to any Environmental Liability to the extent attributable to the ownership or operation of the Business or the Assets by Sellers or their predecessors or Affiliates on or prior to the Closing Date or to events that have occurred, or conditions that existed, on or prior to the Closing Date;
- (f) all Liabilities, obligations and expenses in respect of any product liability with respect to any product manufactured, distributed or caused to be distributed by or on behalf of Sellers on or before the Closing Date;
- (g) all Liabilities, obligations and expenses of Sellers in connection with this Agreement and the Contemplated Transactions, including, without limitation, those expenses to be paid by Sellers pursuant Section 14.3, other than as set forth in Section 3.3; and
- (h) any and all other direct or indirect liabilities, obligations and expenses of Sellers of any nature whatsoever, known or unknown, whether absolute,

Doc#:D\$3:18979.10 25-090

contingent or otherwise, not expressly assumed by Buyer pursuant to Section 2.3.

ARTICLE 3

PURCHASE PRICE

- 3.1 Closing Date Consideration. Subject to the terms and conditions of this Agreement, in reliance upon the representations, warranties, covenants and agreements of Sellers contained herein, and in consideration of the sale, assignment, transfer and delivery of the Assets referred to in Section 2.1, on the Closing Date, (a) Buyer shall (i) assume the Assumed Liabilities and (ii) pay to Seller the Purchase Price and (b) Seller shall be entitled to retain the Deposit.
- 3.2 Allocation of Purchase Price. The Buyer and the Sellers covenant and agree that the Purchase Price shall be allocated among the Assets as set forth on a schedule to be mutually agreed upon by and between Buyer and Sellers before the Closing Date (the "Allocation Schedule"), which allocation and Allocation Schedule will be prepared in good faith in accordance with the requirements of section 1060 of the Code, and the regulations thereunder. The parties agree to reflect the allocation set forth on the Allocation Schedule on their respective forms filed under section 1060 of the Code, and to file all Tax returns and claims and other statements consistent with the Allocation Schedule. The parties further agree that they shall not make any inconsistent written statement or take any inconsistent position on any Tax returns, in any refund claim, or during the course of any Internal Revenue Service or other Tax audit, so long as there exists a reasonable basis in law to maintain the position reflected by such allocation. Each party shall notify the other parties if it receives notice that the Internal Revenue Service proposes any allocation that is different from the allocation set forth on the Allocation Schedule.
- 3.3 Payment of Transfer Taxes. If any Transfer Taxes are imposed, all such Transfer Taxes shall be paid by the party primarily liable therefor, which party shall, at its own expense, prepare and properly file accurate tax returns and other documentation with respect to such Transfer Taxes on a timely basis. The other party shall pay such primarily liable party an amount equal to one-half of all such Transfer Taxes within 10 days after the receipt of a written request by such primarily liable party containing evidence reasonably satisfactory to the other party of the amount of such Transfer Taxes obligated to be paid. Upon receipt by either party of a notice of collection or assessment of an amount of Transfer Taxes that is payable

Doc#:DS3:18979.10 25-090

one-half by Buyer and one-half by Sellers under this Section 3.3, the receiving party shall deliver the same to the other party and Buyer and Sellers shall jointly decide whether to pay or contest such amount.

3.4 Manner of Payment. The Purchase Price shall be paid in immediately available funds by wire transfer to an account designated by Sellers in writing to Buyer not less than two Business Days before the Closing Date.

ARTICLE 4

CLOSING

- 4.1 <u>Time and Place</u>. The closing (the "<u>Closing</u>") of the Contemplated Transactions and the transfer and delivery of all documents and instruments necessary to consummate the Contemplated Transactions shall, subject to the satisfaction of the conditions to Closing, be held at the offices of Much Shelist Freed Denenberg Ament Bell & Rubenstein, P.C. 200 N. LaSalle Street, Suite 2100, Chicago, Illinois 60601-1095, on March 18, 1997, or within three (3) Business Days after the conditions set forth in Sections 9.5 and 10.4 have been satisfied, but in no event later than March 21, 1997, or at such other place or time as Buyer and Sellers may agree in writing (such time and date is herein referred to as the "<u>Closing Date</u>").
- 4.2 <u>Closing Deliveries of Sellers</u>. At the Closing, Sellers shall deliver, or caused to be delivered, to Buyer the following documents, duly executed by or on behalf of Sellers:
 - (a) the Bill of Sale;
 - (b) certified copy of the Sale Order;
- (c) officer's certificate pursuant to Section 9.1;
 - (d) notices pursuant to Section 14.5; and
- (e) such other instruments of conveyance as Buyer shall reasonably request to effect the transfer of the Assets hereunder.
- 4.3 <u>Closing Deliveries of Buyer</u>. At the Closing, Buyer shall pay the Purchase Price and shall deliver, or caused to be delivered, to Sellers the following documents, duly executed by or on behalf of Buyer;
 - (a) the Bill of Sale; and

Doc#:DS3:18979.10 25-090

(b) officer's certificate pursuant to Section 10.1.

ARTICLE 5

REPRESENTATIONS AND WARRANTIES OF SELLERS

Each Seller represents and warrants to Buyer as follows:

- 5.1 Organization of Seller. Seller is a corporation validly existing and in good standing under the laws of the State of Delaware. Seller has all requisite corporate power and authority to own and hold its Assets, and to conduct its Business as heretofore conducted.
- 5.2 Authority to Execute and Perform Agreement. Except as set forth in Schedule 5.2 and subject to and after giving effect to (i) the Approval of the Bankruptcy Court and (ii) compliance with applicable requirements of the Bankruptcy Code, Seller has all requisite corporate power and authority required to enter into, execute and deliver this Agreement, and all other documents contemplated hereby, and to perform fully its obligations hereunder and thereunder. Except as set forth in Schedule 5.2 and subject to and after giving effect to (i) the Approval of the Bankruptcy Court and (ii) compliance with applicable requirements of the Bankruptcy Code, this Agreement has been, and upon execution and delivery by Seller of all other documents contemplated hereby, such documents will be, duly authorized, executed and delivered and constitute (or with respect to all other documents contemplated hereby will constitute) the legal, valid and binding obligations of Seller enforceable against Seller in accordance with their respective terms.
- 5.3 Consents and Approvals; No Violations. Except as set forth on Schedule 5.3 and except for the approval of the Bankruptcy Court under Sections 105, 363 and 365 of the Bankruptcy Code, and except for Approvals under the HSR Act, no filings with, notices to, or Approvals of any Governmental Body or other Person are necessary to be made, given or obtained in connection with the execution and delivery by Seller of this Agreement or any other document contemplated hereby and the consummation and performance by Seller of its obligations hereunder. Subject to Bankruptcy Court Approval and Approvals under the HSR Act, and except as set forth on Schedule 5.3, the execution and delivery of this Agreement, the consummation of the Contemplated Transactions and the performance by Seller of this Agreement will not (i) violate any provision of Seller's articles of incorporation or by-laws; (ii) violate any Order of any

Doc#:DS3:18979.10 25-090

Governmental Body against, or binding upon, Seller or the Assets; or (iii) violate any Law.

- 5.4 <u>Permits</u>. All Permits material to the Business of Seller are set forth on Schedule 5.4.
- 5.5 Litigation. Other than the Chapter 11 Case and except as set forth on Schedule 5.5 and the Bankruptcy Schedules, there are not, to the knowledge of Seller, any Claims (including, without limitation, any Claims relating to the Designated Contracts) or investigations (whether or not the defense thereof or Liabilities in respect thereof are covered by insurance) pending or threatened against or involving any of the Assets which Claim or investigation would have a Material Adverse Effect or impair Seller's right to consummate the Contemplated Transactions, nor is there, to the knowledge of Seller, any Order of any Governmental Body outstanding against Seller or involving any of its Assets which Order would have a Material Adverse Effect or impair Seller's right to consummate the Contemplated Transactions.
- 5.6 <u>Contracts</u>. True and correct copies of all Designated Contracts have been or, prior to February 28, 1997, will be delivered to Buyer.
- 5.7 No Broker. Seller has not retained or utilized the services of any financial advisor, broker or finder in connection with the Contemplated Transactions for which Buyer could be liable for payment of any amount.
- 5.8 <u>Affiliate Assets</u>. All material Affiliate Assets are set forth on Schedule 5.8.
- 5.9 <u>Title to Assets</u>. Upon delivery of and payment for the Assets as herein provided, Buyer will have good and valid title thereto, free and clear of any Liens or Claims, except for (a) Permitted Liens and (b) any Designated Contract as to which a motion pursuant to Section 365 of the Bankruptcy Code is pending on the Closing Date.
- 5.10 Other Assets. Except as otherwise set forth in Schedule 5.10, there are no other material assets owned by Seller or any of its Affiliates used in the Business conducted by Seller that are not included in the Assets or Excluded Assets.
- 5.11 <u>Benefit Plans</u>. There are no other material written Benefit Plans, Employee Benefit Flans or Collective Bargaining Agreements other than those set forth in Schedule 5.11. Solely for the purpose of providing information to Buyer of the type of employee benefits provided to Employees prior to the Closing Date, Sellers shall make available to

Doc#:DS3:18979.10 25-090

Buyer and its Representatives prior to Closing all agreements, summary descriptions and summaries of material modifications or amendments with respect to all Employee Benefit Plans and Collective Bargaining Agreements.

5.12 No Known Breach or Impediment. Seller has no knowledge of any representation or warranty of Buyer herein being untrue or incorrect or any event or circumstance making satisfaction of any condition in Article 9 or 10 unlikely that has not been disclosed in writing to Buyer.

ARTICLE 6

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to Sellers as follows:

- 6.1 Organization of Buyer. Buyer is validly existing and in good standing under the laws of the State of Delaware.
- 6.2 Authority to Execute and Perform Agreement. Buyer has all requisite power and authority required to enter into, execute and deliver this Agreement, and all other documents contemplated hereby, and to perform fully its obligations hereunder and thereunder. This Agreement has been, and upon execution and delivery by Buyer of all other documents contemplated hereby, such documents contemplated will be duly authorized, executed and delivered and constitute the legal, valid and binding obligations of Buyer enforceable against Buyer in accordance with their respective terms.
- 6.3 No Broker. Buyer has not retained or utilized the services of any financial advisor, broker or finder in connection with the Contemplated Transactions for which Sellers could be liable for payment of any amount.
- 6.4 Consents and Approvals. Except for (i) the Approval of the Bankruptcy Court, (ii) any Approvals under the HSR Act and (iii) as set forth on Schedule 6.4, no filings with, notices to, or Approvals of any Governmental Body or Person are necessary to be made, given or obtained in connection with the execution and delivery by Buyer of this Agreement or any other document contemplated hereby and the consummation and performance by Buyer of the Contemplated Transactions, and except as set forth on Schedule 6.4, the execution and delivery of this Agreement, the consummation of the Contemplated Transactions and the performance by Buyer of this Agreement will not (i) violate

Doc#:DS3:18979.10 25-090

any provision of Buyer's organizational documents; (ii) gviolate any Order of any Governmental Body against, or binding upon, Buyer; or (iii) violate any Law.

6.5 No Known Breach or Impediment. Buyer has no knowledge of any representation or warranty of Sellers herein being untrue or incorrect or any event or circumstance making satisfaction of any condition in Article 9 or 10 unlikely that has not been disclosed in writing to Sellers.

ARTICLE 7

CONDUCT PRIOR TO CLOSING

Each Seller covenants and agrees with Buyer that, from and after the date of this Agreement and until the Closing Date or termination of this Agreement (as the case may be), except as otherwise specifically consented to or approved by Buyer in writing or as contemplated by this Agreement:

- 7.1 <u>Conduct of Business</u>. Seller shall take no action respecting its Business outside of the ordinary course, taking into account the Chapter 11 Case. Prior to Closing or termination of this Agreement (as the case may be), Sellers shall not engage in any transaction or take any action that would have a Material Adverse Effect. Nothing in this Section 7.1 shall prohibit Seller, prior to the Closing, from (a) laying off or terminating the employment of any of its employees or (b) selling inventory or trailers in the ordinary course of business, provided that Seller consults in good faith with Buyer before taking any such action. Buyer acknowledges and agrees that (a) Sellers may prior to Closing commence litigation concerning or settle a certain claim by, between and among Sellers, C&C Transportation, Inc. ("C&C") and Banc One Leasing Corporation in the principal amount of approximately \$314,000, which litigation or settlement may result in the delivery to C&C of certain trailers, and (b) Buyer shall have no rights in or to the proceeds collected by Sellers from such litigation or settlement,
- 7.2 <u>Litigation</u>. From the date hereof through the Closing Date or termination of this Agreement (as the case may be), Seller shall promptly notify Buyer of any Claims which after the date hereof are threatened in writing or commenced against Seller or against any consultant, agent, shareholder or other Representative of Seller and which, if adversely determined, would have a Material Adverse Effect.

Doc#:DS3:18979.10 25-090

- 7.3 Access to Information and Employees. the date hereof through the Closing Date or termination of this Agreement (as the case may be), during reasonable business hours and upon reasonable notice, Seller shall use reasonable efforts to make available to Buyer and its Representatives, for purposes of consummating the Contemplated Transactions, all Employees (provided that Seller's president or executive vice president is present), agents and accountants of the Seller and all books, records and other documents concerning the Business, and Buyer shall be entitled, through its Representatives, to make any other reasonable investigation of the Business, provided that no investigation by Buyer shall unreasonably interfere with the conduct of such Business. If Buyer or Seller obtains knowledge of (a) any representation or warranty of the other party being untrue or incorrect in any material respect or any matter which, if existing, occurring and known at the date of this Agreement would have been required to be disclosed in order for its representations or warranties to be true and correct or (b) any event or circumstance making satisfaction of any condition in Article 9 or 10 unlikely, Buyer or Seller (as the case may be) shall promptly notify the other party thereof, in writing. Notwithstanding the foregoing, nothing herein shall require Seller or any of its Affiliates to disclose privileged information, breach any confidentiality or nondisclosure agreement with a third party or commit any violation of Law. No investigation by any party shall diminish, obviate or qualify in any respect any of the representations, warranties, covenants or agreements of any other party under this Agreement; provided, however, that no party may terminate this Agreement on account of the breach of any representation, warranty, covenant or agreement known by such party to have been breached at the time of the Closing. All of the information disclosed to Buyer shall be subject to the Confidentiality Agreement dated October 14, 1996 between Monon and Asset Management Associates of New York, Inc.
- 7.4 <u>Insurance</u>. Seller or its Affiliates shall keep in effect all insurance policies in effect as of the date hereof insuring the Assets and operations of the Business until the end of the day on which the Closing occurs or this Agreement is terminated (as the case may be), unless Buyer shall have earlier obtained appropriate coverage and notified Seller in writing to that effect.
- 7.5 <u>Bankruptcy Filings</u>. Until the Closing occurs or this Agreement is terminated (as the case may be), Seller shall deliver to Buyer copies of all pleadings, motions, notices, statements, schedules, applications, reports and other papers that Seller intends to file in Seller's chapter 11 case within a reasonable time after filing, but

Doc#:DS3:18979.10 25-090

with respect to any such papers that relate to Buyer or the Assets, not less than one (1) Business Day before the filing of such papers. With respect to papers described in this Section 7.5 relating to Buyer, Seller will afford Buyer the opportunity to review and comment regarding all statements contained in such papers as they relate to Buyer.

ARTICLE 8

AGREEMENT OF THE PARTIES AS TO CERTAIN PRE-CLOSING ACTIONS

- Change and Use of Sellers' Names. On or before the Closing Date, Sellers shall (a) take or cause to be taken such action as may be required to change the corporate name of each Seller to a name that is not the same as, or confusingly similar to, such Seller's current corporate name (and shall in any event have deleted the name Monon therefrom), and promptly thereafter Sellers shall deliver to Buyer evidence that whatever filings are necessary in those jurisdictions, if any, in which Sollers are licensed or qualified to do business to effect such name changes have been made, and (b) seek approval of the Bankruptcy Court, pursuant to an application in form and substance reasonably satisfactory to Buyer, to amend the caption of the Chapter 11 Case to reflect the name changes of Sellers. Sellers shall be permitted to use or refer to the current corporate names of Sellers where required by Law.
- 8.2 Filings. Unless Buyer earlier notifies Seller in writing that no such filing is required by law, Buyer and Sellers shall file or cause to be filed with the Federal Trade Commission ("FTC") and the United States Department of Justice ("DOJ") any notifications required to be filed under the HSR Act with respect to the Contemplated Transactions within three (3) Business Days after Sellers' execution of this Agreement, requesting in the notifications an early termination of the HSR Act waiting period. Each party shall use its reasonable efforts to make such filings at the agreed upon time, and to respond, as soon as practicable, to any requests for additional information made by either of such agencies. Any filing fees required by the FTC or DOJ in connection with such notifications will be the responsibility of Buyer.
- 8.3 Access to Leased Property. Prior to the Closing Date or termination of this Agreement (as the case may be), Sellers shall use reasonable efforts to permit Buyer and its Representatives reasonable access to the premises covered by any lease on which either Seller is a tenant, for performance of any reasonable investigations,

Doc#:DS3:18979.10 25-090

provided, that no investigation by Buyer shall unreasonably interfere with the conduct of such Seller's Business. Buyer shall provide reasonable advance notice to Sellers and shall conduct such activities consistent with Sellers' safety policies and procedures applicable to activities conducted at or on the premises covered by any such lease and, to the extent practicable, during Sellers' normal business hours.

8.4 <u>Conditions</u>. Buyer and Sellers agree to use reasonable efforts to fulfill the conditions set forth in Articles 9 and 10 below, respectively.

ARTICLE 9

CONDITIONS PRECEDENT TO THE OBLIGATION OF BUYER TO CLOSE

The obligation of Buyer to consummate the Contemplated Transactions is subject to the fulfillment on or prior to the Closing Date of the following conditions, any one or more of which may be waived by Buyer:

- Representations, Warranties and Covenants. The representations and warranties of Sellers contained in Article 5 shall be true on the Closing Date with the same effect as though made on the Closing Date, except where the breach of such representations and warranties, or any one of them, would not have a Material Adverse Effect. Sellers shall have performed and complied with all covenants or agreements required by this Agreement to be performed or complied with by Sellers in all material respects on or prior to the Closing Date, except where the non-performance or non-compliance of such covenants or agreements would not have a Material Adverse Effect. Each Seller shall have delivered to Buyer a certificate, dated the Closing Date and signed by a duly authorized officer of such Seller, acting in his/her capacity as an officer of such Seller, to the foregoing effect and stating that all conditions to the Buyer's obligations hereunder have been satisfied.
- 9.2 <u>No Orders</u>. On the Closing Date, there shall be no Order of any nature issued by a Governmental Body or other action taken by a Governmental Body, restraining, enjoining or otherwise prohibiting the Contemplated Transactions.
- 9.3 Approvals of Congress Financial. The Approval of Congress Financial necessary for the consummation by Sellers of the Contemplated Transactions and the execution and delivery of this Agreement by Sellers shall have been obtained.

Doc#:DS3:18979.10 25-090

- 9.4 <u>Title to the Assets</u>. Upon consummation of the Contemplated Transactions, Buyer will own and have good and valid title to all of the Assets free and clear of any Liens or Claims, other than (a) Liens created by Buyer or arising from Buyer's ownership of the Assets, (b) Permitted Liens and (c) any Designated Contract as to which a motion pursuant to Section 365 of the Bankruptcy Code is pending on the Closing Date.
- 9.5 <u>Bankruptcy Court Approval</u>. The Bankruptcy Court shall have entered the Sale Order on or before March 21, 1997 and no stay of such Order shall be in effect.
- 9.6 Required Filings/Notices. Unless Buyer has notified Sellers pursuant to Section 8.2 that filing of a notification and report form in compliance with the HSR Act is not required by law, (a) Sellers and their Affiliates shall have filed such notification and report form, and (b) the applicable waiting period with respect to each such form (including any extension thereof by reason of a request for additional information) shall have expired or been terminated.
- 9.7 <u>Financing</u>. (a) Buyer shall have obtained a commitment in writing from its funding source, Congress Financial, sufficient to consummate the Contemplated Transactions on terms satisfactory to Buyer and (b) Congress shall not be in breach of its obligation to provide such funding pursuant thereto.
- 9.8 Rosby Corporation Assets. Buyer shall have entered into a contract with Rosby Corporation, on terms satisfactory to Buyer, for the purchase or license by Buyer of any or all of the assets, properties and rights (including, but not limited to, patents) owned by Rosby Corporation and used in the Business of either Seller ("Affiliate Assets"); provided, however, that the Affiliated Assets shall not include cash of Rosby Corporation or the stock of either Monon or MTI of Detroit, Inc.

Doc#:DS3:18979.10 25-090

9.9 Minimum Inventory Level. On or before the date on which Competing Bids were due pursuant to the Overbidding Procedures (the "Competing Bid Date"), Buyer shall have completed its investigation of Sellers' inventory levels and shall be satisfied that the total amount of Sellers' inventories on hand as of such date, at cost and without adjustments, was at least \$11,000,000. Unless Buyer shall give notice to Sellers (with a copy thereof being sent by Buyer by facsimile transmission to Congress Financial, c/o Latham & Watkins, Attention: Douglas Bacon, Esq., Telecopier No. 312-993-9767) by 5 p.m.(Central Time) on the Competing Bid Date that this condition has not been satisfied, such condition shall be deemed to have been waived.

ARTICLE 10

CONDITIONS PRECEDENT TO THE OBLIGATION OF SELLERS TO CLOSE

The obligation of Sellers to consummate the Contemplated Transactions is subject to the fulfillment on or prior to the Closing Date of the following conditions, any one or more of which may be waived by Sellers:

- The representations and warranties of Buyer contained in this Agreement shall be true in all material respects on the Closing Date with the same effect as though made on the Closing Date. Buyer shall have performed and complied with all covenants and agreements required by this Agreement to be performed or complied with by it in all material respects on or prior to the Closing Date. Buyer shall have delivered to Sellers a certificate, dated the Closing Date and signed by a duly authorized representative of Buyer, acting in his/her capacity as a representative of Buyer, to the foregoing effect and stating that all conditions to Sellers' obligations hereunder have been satisfied.
- 10.2 <u>No Orders</u>. On the Closing Date, there shall be no Order of any nature issued by a Governmental Body or other action taken by a Governmental Body, restraining, enjoining or otherwise prohibiting the Contemplated Transactions.
- 10.3 <u>Approvals</u>. All Approvals necessary for the consummation of the Contemplated Transactions and the execution and delivery of this Agreement by Sellers shall have been obtained.

Doc#: DS3:18979.10 25-090

- 10.4 <u>Bankruptcy Court Approval</u>. The Sale Order shall have been entered and no stay of such Order shall be in effect.
- 10.5 HSR Filings. (a) Buyer and its Affiliates, if required in connection with the Contemplated Transactions to file a notification and report form in compliance with the HSR Act, shall have filed such forms, and (b) the applicable waiting period with respect to each such form (including any extension thereof by reason of a request for additional information) shall have expired or been terminated.
- 10.6 <u>Purchase Price</u>. Buyer shall have paid the Purchase Price.
- 10.7 <u>Assumption of Liabilities</u>. Buyer shall have assumed (a) the Assumed Liabilities pursuant to the Bill of Sale and (b) the Congress Debt.
- 10.8 <u>Commitment from Congress Financial</u>. Buyer shall have either (a) obtained a commitment from Congress Financial for the funding referred to in Section 9.7 or (b) waived said condition in writing, in either case within three (3) Business Days after Sellers' execution of this Agreement.
- 10.9 <u>Creditors Committee Approval</u>. The Official Committee of Unsecured Creditors appointed in the Chapter 11 Case shall have consented to the Contemplated Transactions.
- 10.10 <u>Completion of Auction</u>. The competitive bidding process contemplated by Sellers' application for approval of the Sale Order has been concluded and Buyer is the successful bidder.
- 10.11 Rosby Corporation Assets. Buyer shall have either (a) entered into the contract with Rosby Corporation referred to in Section 9.8 or (b) waived said condition in writing, in either case within five (5) Business Days of the filing of the application referred to in Section 11.1.

ARTICLE 11

BANKRUPTCY COURT APPROVALS; OVERBIDDING PROVISION; BREAK-UP FEE AND EXPENSE REIMBURSEMENT

11.1 <u>Sale Order</u>. Sellers' obligations hereunder are subject to approval of the Bankruptcy Court. Upon Sellers' execution of this Agreement, Sellers shall promptly seek, and shall use their best efforts to obtain, approval by the Bankruptcy Court, under Sections 105, 363 and 365 of the Bankruptcy Code, of the Contemplated Transactions,

Doc#:DS3:18979.10 25-090

including the procedures set forth in Sections 11.4 and 11.5 of this Agreement, pursuant to (a) an application in form and substance reasonably acceptable to Buyer and (b) an Order substantially in the form annexed hereto as Exhibit C (the "Sale Order"). Buyer shall cooperate with Sellers to obtain such approval.

- 11.2 Failure to Obtain Approval. Sellers or Buyer may terminate their respective obligations to sell or acquire the Assets if the Bankruptcy Court has not entered the Sale Order by March 21, 1997.
- Competing Bid. If, prior to the Closing 11.3 Date or termination of this Agreement (as the case may be), a proposal or offer (a "Competing Bid") from any Person or Persons (a "Competing Bidder") relating to a transaction that is substantially equivalent (except as to price) to the transactions as provided for in this Agreement, or relating to an alternative transaction or transactions consisting of any acquisition or purchase of all or substantially all of the Assets, or any equity interest in Sellers, or any -merger, consolidation, reorganization, recapitalization-or- business combination with Sellers or any other transaction having reasonably similar effect, is presented to Sellers or the Bankruptcy Court, which in the reasonable and good faith judgment of Sellers provides greater value to Sellers, then Sellers shall promptly communicate to Buyer the terms of the Competing Bid.
- 11.4 Overbidding Procedures. The overbidding procedures shall be as set forth on Exhibit \underline{D} hereto.

11.5 Payment of Break-Up Expenses and Fee.

(a) If (i) as of the date of the hearing on the application referred to in Section 11.1 (A) Buyer has not committed any breach of this Agreement, (B) neither Buyer nor Sellers have exercised their right to terminate this Agreement, and (C) Buyer confirms in writing that it is not aware of any condition to Sellers consummating the Closing (other than Section 10.4) which has not been satisfied or which, if not satisfied and in Buyer's control, cannot be satisfied by Buyer, (ii) a Competing Bid from a Person other than Buyer or any Affiliate of Buyer (whether or not conforming with the provisions of Section 11.4) is approved by Order of the Bankruptcy Court and consummated and (iii) Buyer remains committed, and is able, to consummate the Contemplated Transactions, subject to satisfaction (or waiver) of the conditions set forth in Article 9, if consummation of the transactions contemplated by the Competing Bid does not occur, Buyer shall be paid simultaneously with the consummation of the transactions contemplated by the Competing Bid, an amount equal to

Doc#:DS3:18979.10 25-090

- (i) the reasonable and documented out-of-pocket costs and expenses (including, without limitation, attorneys', financial advisors', accountants', engineers' and other consultants' fees) incurred by Buyer, not to exceed \$150,000 in the aggregate (the "Break-Up Expenses"), in connection with (A) the preparation, negotiation and execution of the term sheet and this Agreement, (B) the due diligence efforts of Buyer and its professionals and advisors in connection with the Contemplated Transactions, (C) the financing of the Contemplated Transactions, (C) the financing of the pursuit of the Contemplated Transactions, plus (ii) \$200,000 (the "Break-Up Fee"), plus the cost of any filing fees paid by Buyer pursuant to Section 8.2.
- (b) Payment of the Break-Up Expenses and Break-Up Fee shall be full consideration for the Buyer's efforts and expenses in connection with the term sheet, this Agreement and the Contemplated Transactions, including the substantial due diligence efforts of Buyer and its professionals and advisors, and Sellers shall have no further liability therefor.
- (c) The Break-Up Expenses and Break-Up Fee, if not paid from another source, will constitute an administrative expense of Sellers pursuant to Section 503(b) of the Bankruptcy Code and a first priority administrative expense (an "Administrative Expense") payable from the sales proceeds of any Competing Bid.

ARTICLE 12

TERMINATION OF AGREEMENT

- 12.1 <u>Termination</u>. This Agreement may be terminated prior to the Closing by giving written notice of termination after the event or circumstance giving rise to such right of termination as follows:
- (a) at any time on or prior to the Closing Date, by mutual written consent of Buyer and Sellers;
- (b) at the election of Sellers, if any one or more of the conditions to the obligations of Sellers to close as set forth in Article 10 has not been fulfilled on the Closing Date or if any of those conditions have not been satisfied on a timely basis;
- (c) at the election of Buyer, if any one or more of the conditions to the obligations of Buyer to close as set forth in Article 9 has not been fulfilled on the Closing Date or if any of those conditions have not been satisfied on a timely basis;

Doc#:DS3:18979.10 25-090

- (d) by either Buyer or Sellers, if any Governmental Body shall have issued an Order, or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated hereby (which the party seeking to terminate this Agreement shall have used all reasonable efforts to have lifted or reversed) and such Order shall have become final and nonappealable;
- (e) by either Buyer or Sellers if the Bankruptcy Court has not entered the Sale Order by March 21, 1997;
- (f) automatically, upon consummation of the transactions contemplated by a Competing Bid, in which case the provisions of Section 11.5 shall survive and be applicable; and
- (g) by either Buyer or Sellers, if the parties are unable to reach agreement within five (5) Business Days after Sellers' execution of this Agreement regarding any material exhibit or schedule to this Agreement that is not attached hereto at the time of Sellers' execution of this Agreement, but Buyer and Sellers agree to act in good faith and use their best efforts to reach agreement on all such exhibits and schedules as promptly as possible after execution of this Agreement.

12.2. Remedies in the Event of Failure to Close.

- (a) Upon termination of this Agreement, Sellers shall promptly return the Deposit to Buyer, except that Sellers may retain the Deposit as liquidated damages if, and only if, at the time of such termination Buyer is in material breach of any of its representations, warranties, covenants, agreements or obligations contained herein and neither Seller is in material breach of this Agreement.
- (b) Upon termination of this Agreement pursuant to Section 12.1(c) after entry of the Sale Order by the Bankruptcy Court by reason of either Seller's willful action, Sellers, in addition to returning the Deposit to Buyer, shall pay to Buyer as liquidated damages the Break-Up Expenses as an Administrative Expense if, and only if, (i) at the time of such termination all of the conditions to Sellers' obligation to consummate the Closing have been satisfied (other than the conditions that would be satisfied upon payment of the Purchase Price and delivery of the documents referred to in Section 4.3) and (ii) Buyer confirms in writing to Sellers that Buyer is ready, willing and able to pay the Purchase Price, deliver the documents referred to in Section 4.3 and consummate the Closing.

Doc#:DS3:18979.10 25-090

- (c) Upon termination of this Agreement for any reason, neither Buyer nor Sellers shall be liable for any payment or shall have any other obligation or liability whatsoever hereunder, except as expressly provided in Section 12.2(a), Section 12.2(b) or Article 11, if applicable.
- (d) This Article 12 sets forth the sole and exclusive rights and remedies of the parties in the event of a breach of a representation, warranty, covenant or agreement contained in this Agreement, including, without limitation, any failure to consummate the Closing (except as provided in Article 11, if applicable).
- 12.3 Effect of Termination. If this Agreement is terminated and the transactions contemplated hereby are not consummated, this Agreement shall become void and of no further force and effect, except that any such termination shall be without prejudice to the rights and obligations of the parties hereto under Sections 11.5 and 12.2(a) and (b).

ARTICLE 13

CONDITION OF ASSETS

At the Closing, subject to the terms and conditions of this Agreement, Buyer shall accept the condition of the Assets and the Assumed Liabilities "AS IS, WHERE IS," without any representation, warranty or guarantee, express or implied, as to merchantability, fitness for a particular purpose or otherwise as to condition, size, extent, quantity, quality, type or value. No representation or warranty herein by Sellers shall survive Closing or termination of this Agreement, as the case may be, provided, however, that Buyer shall not be deemed to have waived any claims against Sellers for (i) fraud in connection with this Agreement or the Contemplated Transactions or (ii) failure to deliver the Assets to Buyer with good and valid title to the Assets free and clear of any Claims and Liens, except for (x) Liens created by Buyer or arising out of Buyer's ownership of the Assets or (y) Permitted Liens.

ARTICLE 14

MISCELLANEOUS

14.1 <u>Certain Tax Matters</u>. Notwithstanding anything herein to the contrary, Buyer and Sellers shall each permit Representatives of the other, at all reasonable times after the Closing and upon reasonable notice, access

Doc#:DS3:18979.10 25-090

to their offices, records and accounts which relate to Sellers or the Assets for the purposes of obtaining any available information necessary for any reasonable business purpose, including, but not limited to, the preparation and filing of any Tax returns and other reports to any Governmental Body for any period.

- 14.2 Employees. Within ten (10) Business Days after their execution of this Agreement, Sellers shall deliver to Buyer a true and complete list of all Employees of Sellers. As soon as reasonably practicable after receipt of Sellers' list, Buyer shall deliver to Sellers a list of the Employees to which Buyer has extended offers of employment, disclosing when such employment will become effective.
- 14.3 Expenses. Except as contemplated by Sections 3.3, 11.5, 12.2(b), 14.7 and 14.9, all expenses of the preparation, execution and consummation of this Agreement and of the Contemplated Transactions, including, without limitation, attorneys', accountants' and outside advisors' fees and disbursements, shall be borne by the party incurring such expenses.
- 14.4 Collections. After the Closing Date, Buyer shall have the sole right and authority to collect, for its own account and sole benefit, all monies payable in respect of the Assets, no matter how or when earned (including all Assets or Accounts Receivable in existence on the Closing Date). If Sellers (or their successors) or any of their Affiliates shall receive any such monies, they shall hold all such monies for the sole benefit of Buyer. Within three (3) Business Days after receipt thereof, Sellers (or their successors) shall cause the transfer and delivery to Buyer of any monies or other property which Sellers (or their successors) may receive after the Closing Date in payment of amounts payable in respect of the Assets or Accounts Receivable. Sellers authorize Buyer to endorse in either Seller's name all notes, checks, drafts, money orders or other instruments of payment in respect of the foregoing which may come into the possession of Buyer, and Sellers hereby ratify all that Buyer shall lawfully do or cause to be done by virtue hereof. This right shall become irrevocable upon Closing. If any monies turned over to Buyer pursuant to this Section 14.4 are not collected by Sellers (e.g. a check returned for insufficient funds) or are required to be returned by Sellers, Buyer shall pay back to Sellers such amount upon receipt of evidence reasonably satisfactory to Buyer that Sellers have not received or are required to return such monies. If for any reason the amounts returned by Buyer to Sellers are later recouped by Sellers, Sellers shall as soon as reasonably practicable pay over such amount to Buyer.

Doc#:DS3:18979.10 25-090

- 14.5 Notice to Third Parties. On the Closing Date, Sellers shall execute and deliver to Buyer for use by Buyer with respect to Persons, such as licensors, licensees, sub-licensees and any other third party, that currently have any rights whatsoever in or to the Assets, anywhere throughout the world, letters prepared jointly by, and mutually acceptable to, Buyer and Sellers (a) advising that (i) effective as at the Closing Date, Buyer owns Sellers' interests in or to the Assets; (ii) Buyer is entitled to receive all moneys, payments, receipts, revenues or income derived therefrom in accordance with this Agreement (including all Accounts Receivable in existence on the Closing Date); (b) requesting that thereafter such Persons render all required statements and activity reports and make payments of one hundred percent (100%) of all sums thereafter otherwise due in respect of any period, whether before or after Closing, to Buyer in accordance with this Agreement, and provide copies of all notices, claims or other correspondence, directly to Buyer and (c) advising that Buyer has only assumed the Assumed Liabilities and that Sellers have retained, without affecting the applicability of the bankruptcy Code to their rights and obligations with respect to third parties, all of their other Liabilities.
- 14.6 <u>Documents and Other Information</u>. After the Closing Date, Sellers agree to make available to Buyer, and to cause their Affiliates to make available to Buyer, such documents and other information used by Sellers in their Businesses, which documents and information are not being purchased by Buyer pursuant to this Agreement. Sellers further agree to notify Buyer if any Seller or any of its Affiliates wishes to destroy any documents or information used in such Seller's Business, and to provide Buyer with a reasonable opportunity to recover and retain any such documents or information scheduled to be destroyed.
- 14.7 Dispute Resolution. The parties agree that all disputes under this Agreement shall be submitted to the Bankruptcy Court, and the determination of the Bankruptcy Court shall be final and binding on all parties in interest, and each of the parties hereby waives any rights of appeal or review which it may otherwise have had in connection with such determination. Additionally, the parties agree that the prevailing party in a Bankruptcy Court proceeding which was the subject of a dispute under this Agreement shall be entitled to recover all reasonable counsel fees and expenses incurred in connection with such dispute in the Bankruptcy Court. The amount of such counsel fees and costs, and any interest on amounts in dispute, if not agreed to between or among the parties, shall be subject to final determination of the Bankruptcy Court, and each of the parties hereby waives any rights of appeal or review which it may otherwise have had in connection with such determination.

Doc#:DS3:18979.10 25-090

14.8 <u>Severability</u>. In case any provision of this Agreement shall be deemed to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby.

14.9 Further Assurances.

- (a) Sellers and Buyer agree to execute and deliver, or to cause to be executed and delivered, all such instruments, and to take all such action, as the other may reasonably request in order to effectuate the intent and purposes of, and to carry out the terms of, this Agreement, including, without limitation, the execution of any additional assignments or other documentation prepared by Buyer that are necessary for Buyer to evidence or record the transfer to Buyer of the Assets in accordance with this Agreement; provided, however, that Sellers shall not be required to take any such action not otherwise required of them by the provisions of this Agreement unless Buyer agrees to reimburse them for their expenses.
- (b) Without limiting the generality of the foregoing, at any time and from time to time after the Closing Date, at the request of Buyer and without further consideration, Sellers shall execute and deliver such other instruments of transfer, conveyance, assignment and confirmation and shall take such action, including making application to the Bankruptcy Court, as may be necessary to effectively transfer, convey and assign to Buyer, and/or to confirm Buyer's title to, all of the Assets, or to put Buyer in actual possession and operating control thereof, and/or to permit Buyer to exercise all rights with respect thereto (including, without limitation, rights under Designated Contracts and other arrangements as to which the Approval of any third party to the transfer thereof shall not have previously been obtained); provided, however, that Sellers shall not be required to take any such action unless Buyer agrees to reimburse Sellers for their expenses.
- (c) Buyer understands that Sellers are selling all of their facilities and substantially all of their assets pursuant to this Agreement, and expect to terminate the employment of most of their employees. In addition, Buyer understands that Sellers are in the process of winding up their affairs and will not remain in existence indefinitely. Accordingly, Sellers' obligations under this Section 14.9, and any other obligations of Sellers that extend beyond the Closing Date, are subject to the foregoing.

Doc#:DS3:18979.10 25-090

14.10 Notices. Any notice or other communication required or permitted hereunder shall be in writing and shall be delivered personally, telegraphed, sent by facsimile transmission, overnight delivery service or certified, registered or express mail, postage prepaid. Any such notice shall be deemed given when so delivered personally, telegraphed or sent by facsimile transmission, or if by overnight delivery service, one day after the date of deposit to such overnight delivery service or, if mailed, three (3) Business Days after the date of deposit in the United States mail, as follows:

(i) if to Buyer, to:

HPA Asset, Inc. c/o Asset Management Associates 117 Whites Lane Southampton, N.Y. 11968 Attn: Charles S. Holmes

Telephone: (201) 531-9250 Telecopier: (201) 531-9249

with a copy to:

Paul, Weiss, Rifkind, Wharton & Garrison 1285 Avenue of the Americas New York, New York 10019-6064 Attention: Robert L. Laufer, Esq.

Telephone: (212) 373-3333 Telecopier: (212) 373-2053

(ii) if to Sellers, to:

Monon Corp.
One Water Tower Drive
Monon, Indiana 47959
Attention: Thomas J. Rosby

Telephone: (219) 253-6621 Telecopier: (219) 253-8033

with a copy to:

Much Shelist Freed Denemberg Ament Bell & Rubenstein, PC
200 N. LaSalle Street, Suite 2100
Chicago, Illinois 60601
Attention: Larry Brenman, Esq.

Telephone: (312) 621-1471 Telecopier: (312) 621-1750 Any party may by notice given in accordance with this Section 14.10 to the other parties designate another address or Person for receipt of notices hereunder.

- 14.11 Entire Agreement. This Agreement (including the Exhibits and Schedules hereto) contains the entire agreement among the parties with respect to the Assets and the Contemplated Transactions, and supersedes all prior agreements, written or oral, with respect thereto.
- 14.12 Waivers and Amendments: Non-Contractual Remedies. This Agreement may be amended, superseded, canceled, renewed or extended, and the terms and conditions hereof may be waived, only by a written instrument signed by the parties or, in the case of a waiver, by the party waiving compliance. No delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any waiver on the part of any party of any such right, power or privilege hereunder, nor any single or partial exercise of any such right, power or privilege hereunder, preclude any other or further exercise thereof or the exercise of any other such right, power or privilege hereunder. Except as provided herein, the rights and remedies herein provided are cumulative and are not exclusive of any rights or remedies that any party may otherwise have at law or in equity. Nothing contained in this Agreement shall be construed as an admission or waiver by, or estoppel or evidence against, Buyer or any of its Affiliates or Sellers or any of their Affiliates concerning any matter.
- 14.13 Governing Law; Jurisdiction of Court. This Agreement shall be governed and construed in accordance with the laws of the State of Indiana applicable to agreements made and to be performed entirely within such state and, to the extent applicable, the Bankruptcy Code. All disputes arising out of or related to this Agreement, including, without limitation, any dispute relating to the interpretation, meaning or effect of any provision hereof, will be resolved in the Bankruptcy Court and the parties hereto each submit to the exclusive jurisdiction of the Bankruptcy Court for the purpose of adjudicating any such dispute, to the extent that the jurisdiction of the Bankruptcy Court is available.
- 14.14 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument.
- 14.15 <u>Binding Effect; No Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted

Doc#: DS3:18979.10 25-090

assigns, including any person appointed for or in connection with the Chapter 11 Case or in any subsequent case under the Bankruptcy Code in which either Seller may be a debtor. Except as provided in the preceding sentence, this Agreement and the rights and remedies hereunder are not assignable by Sellers or Buyer, except Buyer may assign this Agreement and/or its rights and remedies hereunder (i) to HPA Asset, Inc., a corporation to be formed by the principals of Buyer, or (ii) to any one or more of Buyer's Affiliates or (iii) after Closing in connection with a sale, transfer or other disposition of all, or a portion of, the Assets.

- 14.16 Third Party Beneficiary. No person who is not a party to this Agreement (including, without limitation, any Person who may be deemed to be an incidental beneficiary of any provision of this Agreement) shall be deemed to be a beneficiary of any provision of this Agreement or entitled to rely upon any representation, warranty, covenant or other agreement herein or have any Claim, cause of action, right or remedy pursuant to this Agreement.
- 14.17 <u>Variations in Pronouns</u>. All pronouns and any variations thereof refer to the masculine, feminine or neuter, singular or plural, as the context may require. Any term defined herein may be used in either its singular or plural form whether or not defined in such form.
- 14.18 Exhibits and Schedules. The Exhibits and Schedules are part of this Agreement as if fully set forth herein. All references herein to Sections, subsections, clauses, Exhibits and Schedules shall be deemed references to such parts of this Agreement, unless the context shall otherwise require. The disclosure of any item pursuant to this Agreement, on any Schedule hereto or on the Bankruptcy Schedules shall be deemed to be disclosed for all purposes.
- 14.19 <u>Headings and Table of Contents</u>. The headings and table of contents in this Agreement are for reference only, and shall not affect the interpretation of this Agreement.

Doc#:DS3:18979.10 25-090

IN WINTERS WHEREBY, the parties have executed this Agreement on the data first above written.

HOMAS IT ROSAY

Title: /

THOMAS J. ROSEY PRESIDENT (COO

EFA ASSET, LLC

Titlet

Re-ADMINISTRATE MAN

TOTAL PACE.39 ==

%0 0000 S \$

EEB 54 1664 IS:015W

IN WITNESS WHEREOF, the parties have executed this Agreement on the date first above written.

MONON CORPORATION

By:	
Name: Title:	
MONON TRAILER,	INC.
By: Name: Title:	

HPA ASSET, LLC

Name: CharlesS, Holmes
Title: Hember

Door DS3:18979.10 2534

BILL OF SALE AND ASSIGNMENT AND INSTRUMENT OF ASSUMPTION

THIS BILL OF SALE AND ASSIGNMENT AND INSTRUMENT OF ASSUMPTION (the "Bill of Sale") is dated as of the ____day of March, 1997, by and among HPA ASSET, LLC, a Delaware limited liability company (the "Buyer"), and MONON CORPORATION, a Delaware corporation ("Monon"), and MONON TRAILER, INC., a Delaware corporation ("MTI"), as debtors and debtors in possession (each a "Seller" and, collectively, the "Sellers").

WHEREAS, on September 25, 1996 ("Petition Date"), certain creditors of Monon commenced Case No. 96-40623 by filing an involuntary petition under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101 et sec. (the "Bankruptcy Code"), in the United States Bankruptcy Court for the Northern District of Indiana, and on October 11, 1996, Monon consented to chapter 11 relief, and on the same date, MTI commenced Case No. 96-12098 by filing a voluntary petition under chapter 11 of the Bankruptcy Code, which case is being jointly administered with Case No. 96-40623;

WHEREAS, the Buyer and the Sellers entered into a Purchase Agreement dated as of February 20, 1997 (the "Purchase Agreement"; any capitalized terms used herein and not otherwise defined herein shall have the meaning ascribed to such term in the Purchase Agreement);

WHEREAS, pursuant to the Purchase Agreement, the Sellers will sell and assign to the Buyer, and the Buyer will purchase from the Sellers, all of the Sellers' right, title and interest in and to the Assets upon the terms and subject to the conditions of such Agreement; and

WHEREAS, by this Bill of Sale, the Sellers intend to sell, assign, transfer and deliver to the Buyer all of the properties, assets, rights and interests of the Sellers to be sold, assigned, transferred and delivered to the Buyer under the Purchase Agreement.

NOW, THEREFORE, pursuant to the Purchase Agreement and in consideration of the Purchase Price and of other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and of the mutual covenants set forth herein, the Sellers hereby sell, assign, transfer and deliver to the Buyer, free and clear of any and all Liens and Claims, except Permitted Liens and the Buyer hereby purchases, acquires and takes assignment and delivery of all right, title and interest of the Sellers, in and to the Assets, but provided that, notwithstanding anything in this Bill of Sale to the contrary, there shall be excluded from the Assets for all purposes, the Excluded Assets.

TO HAVE AND TO HOLD all of the aforesaid property, assets, rights and interests hereby given, granted,

Doc#: DS3: 433738.2 2534

conveyed, contributed, assigned, transferred, remised, released, alienated, set over, and confirmed or intended so to be unto the Buyer to and for the use of the Buyer, its successors and assigns, forever.

AND IN FURTHER CONSIDERATION, the Buyer and the the Sellers, for themselves and their respective permitted successors and assigns, have covenanted, bargained and agreed, and by this Bill of Sale covenant, bargain and agree, that, subject to the terms and conditions set forth in the Purchase Agreement and pursuant to Sections 105, 363 and 365 of the Bankruptcy Code, the Buyer hereby assumes and agrees to perform in accordance with their respective terms, (i) the Designated Contracts, including payment of amounts required to cure any defaults necessary for the assumption and assignment thereof pursuant to Section 365(b)(1) of the Bankruptcy Code, whether such amounts accrued before or after the Petition Date, (ii) the Congress Debt and (iii) the Property Taxes.

Except as otherwise provided in the Purchase Agreement, the Buyer expressly does not assume, and the the Sellers agree that, without affecting the applicability of the Bankruptcy Code to their rights and obligation with respect to third parties, the Sellers shall be liable for and shall retain the Excluded Liabilities.

Without limiting anything in the Purchase Agreement, IT IS EXPRESSLY UNDERSTOOD AND AGREED THAT THE BUYER SHALL ACCEPT THE CONDITION OF THE ASSETS AND THE ASSUMED LIABILITIES "AS IS, WHERE IS" WITHOUT ANY REPRESENTATION, WARRANTY OR GUARANTEE, EXPRESS OR IMPLIED, AS TO MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR OTHERWISE OR AS TO THE CONDITION, SIZE, EXTENT, QUANTITY, QUALITY, TYPE, VALUE OF SUCH PROPERTY OR BUSINESS. REPRESENTATION OR WARRANTY SHALL SURVIVE CLOSING OR TERMINATION OF THE PURCHASE AGREEMENT, AS THE CASE MAY BE, PROVIDED, HOWEVER, THAT THE BUYER SHALL NOT BE DEEMED TO HAVE WAIVED ANY CLAIMS AGAINST THE SELLERS FOR (i) FRAUD IN CONNECTION WITH THE PURCHASE AGREEMENT OR THE CONTEMPLATED TRANSACTIONS OR (ii) FAILURE TO DELIVER THE ASSETS TO THE BUYER WITH GOOD AND VALID TITLE TO THE ASSETS FREE AND CLEAR OF ANY LIENS EXCEPT FOR LIENS CREATED BY THE BUYER OR ARISING OUT OF THE BUYER'S OWNERSHIP OF THE ASSETS, AND PERMITTED LIENS; PROVIDED, HOWEVER, THAT ANY CLAIMS MADE BY THE BUYER AGAINST THE SELLER FOR FAILURE TO DELIVER THE ASSETS, OR ANY ONE OF THEM, FREE AND CLEAR OF SUCH LIENS, SHALL BE LIMITED TO THE AMOUNT PAID BY THE BUYER TO THE SELLER FOR THE ASSETS OR ASSET, AS THE CASE MAY BE, WHICH HAS NOT BEEN DELIVERED TO THE BUYER FREE AND CLEAR OF SUCH LIENS.

This Bill of Sale shall be governed by and construed in accordance with the laws of the state of

Doc#:DS3:433738.2 2534

Indiana without regard to such state's conflicts of law principles and shall bind and inure to the benefit of the parties hereto and their respective successors and assigns.

IN WITNESS WHEREOF, the parties have duly executed this Bill of Sale and Assignment and Instrument of Assumption as of the date first above written.

HPA ASSET, LLC
By: Name: Title:
MONON CORPORATION
By:
MONON TRAILER, INC.
By: Name: Title:

Doc#:DS3:433738.2 2534

Exhibit B

PERMITTED LIENS

- 1. Real Estate taxes for 1995 in an amount not exceeding \$170,000 and all real estate taxes for 1996 and subsequent years.
- 2. Exceptions 2-13 (both inclusive), 15, 16, 19, 42-86 (both inclusive), and 89-93 (both inclusive) as set forth on the attached pages 14-21 of Schedule B (which is taken from Ticor Title Insurance Company commitment number 9511-75 dated March 11, 1997 8:00 a.m.).
- 3. Liens in favor of Congress Financial Corporation (Central).
- 4. Any amounts required to cure any defaults necessary for the assumption of Designated Contracts.

Schedule B Continued:

Commitment No.: 9511-75

Parcel No. 020-00013-96 ADDED ASSESSMENT '95 PAY '96 PINE ST BALLS ADD L31

	\$72.	
	\$72.	
Penuity	· · · · · · · · · · · · · · · · · · ·	DIAT TON 85
	Parcel No. 020-88400-00 VACATED PT 1st ST .796	

Land Value - \$2000. Improvements - \$0.00

	i.	ind	Valu	e • 5	4030).	lm	μευ	YUS	ונטח	ís -	\$11.	00					
May 1996 Installment	 			٠.							. ,			 	 	 	\$146.61	NOT PAID
Penalty	 										٠.			 	 	 	\$ 14.66	NOT PAID
Nov 1996 Installment	 			٠.									٠.	 	 	 	\$146.61	NOT PAID
Penalty	 													 	 	 	\$ 14.66	NOT PAID

Parcel Na. 020-99470-00 VACATED FORTION 1st ST .234AC Land Value - \$270. Improvements - \$0.00

May 1996 his allment	\$9.83 NOT PAID
Penulty	\$.98 NOT PAID
Nov 1996 Installment	\$9.83 NOT PAID
Penalty	S.98 NOT PAID

Parcel No. 020-983(0-00 OUT E SE 21-28-4 2.197A Land Value - \$11,130. Improvements - \$8.00

May 1996 Installment	 	٠.	٠.	 	 	 	 	 		 		٠.				 \$404.89	NOT	' PAII)
Penulty	 			 . ,		 	 	 		 						 \$ 40.49	TOM	PAT)
Nov 1996 Installment	 			 		 	 	 		 	٠.			 		 \$404,89	NOT	PAII)
Penulty	 			 		 	 	 		 				 		 \$ 40.49	NOT	PAII)

Taxes for the year 1996 due and physible in 1997 became a lien March 1, 1996, but have not yet been determined, and are not yet due and physible, and all subsequent taxes.

- Any future Assessments for the Harry Kester Maintenance or Reconstruction Ditches. NONE SHOWN DUE AND PAYABLE AT THIS TIME.
- 3. If Kisder & Rr #1 Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monon Corporation at Parcel No. 658-01000-18 over OUT S/S SE SE 18-28-4 25AC shows Total Annual Assessment of \$25.00. Prior Delinquency in the amount of \$25.00 + Penalty of \$2.50 NOT PAID. May 1997 Installment of \$25.00 NOT PAID. Subject to any and all future assessments not yet due and payable.
- 4. H. Kesler & BR #1 Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monon Corporation at Parcel No. 658-01000-43 over 17-28-4 OUT SW SW 11.94 OUT SW SW .78 OUT SW SW .62 shows Total Annual Assessment of \$13.34. Prior Delinquency in the amount of \$13.34 + Penalty of \$1.33. NOT PAID. May 1997 Installment of \$13.34 NOT PAID. Subject to any and all future assessments not yet due and payable.

Schedule B: Page 14 of 21

No state of this

Marily 1997 - 31119M - MUCH SHELIST FDABLE

Schedula B Continued:

Commitment No.: 9511-75

- 5. II. Kesler & BR #1 Ditch entered in the Office of the Tressurer in Ditch Record 1997 Maintenance versus Monan Corporation at Parcel No. 658-01000-42 over OUT SW SW 17-25-4 10.46 shows Total Annual Assessment of \$10.46. Prior Delinquency in the amount of \$10.46 + Penalty of \$1.05 NOT PAID. May 1997 Installment of \$10.46 NOT PAID. Subject to any and all future assessments not yet due and ptyphic.
- 6. II. Kesler & BR #1 Ditch entered in the Office of the Tressurer in Ditch Record 1997 Mainferance versus Monon Corporation at Parcel No. 638-01000-47 over OUT SW COL NW SW 17-28-4 .25 shows Total Annual Assessment of \$5.00. Prior Delinquency in the amount of \$5.00 + Penalty (\$5.50 NOT PAID. May 1997 Installment of \$10.46 NOT PAID. Subject to any and all future assessments out yet due and payable.
- 7. Pern McKillip Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monan Corporation at Parcel No. 628-01002-00 over E NIV 27-28-4 52.85A show: NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due and payable.
- 8. Fern McKillip Ditch entered in the Office of the Treasurer in Office Record 1997 Maintenance versus Monon Corporation at Parcel No. 628-01001-00 over OUT W NE 28-28-4 .75A shows NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due and payable.
- 9. Form McKillip Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monon Corporation at Parcel No. 628-01001-59 aver OUT S/S SE SE 18-28-4 25A shows NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due and payable.
- 10. Fern McKillip Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monon Corporation at Parcel No. 628-01001-82 over OUT SW SW 22-28-4 1.57.. & OUT SW SW 17-28-4 10.46A shows NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due and payable.
- Fern McIGlip Ditch entered in the Office of the Treisurer in Ditch Record 1997 Maintenance versus Monon Trailer, Inc. 21 Parcel No. 628-01003-86 over W NW 27-28-4 77.94A, OUT SW COR E N 27-28-4,73A, PT SW SW 22-28-4,35A shows NONE DUE AND PAYABLE AT TIDS TIME. Subject to any and all future assessments not yet due and payable.
- 12. Fern MclGillip Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance versus Monon Trailer, Inc. at Parcel No. 628-01004-94 over ENE 28-28-4 80.65A sh avs NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due und payable.
- 13. Fern McKillip Ditch entered in the Office of the Treasurer in Ditch Record 1997 Maintenance years Monon Corporation at Parcel No. 628-01003-31 over OUT SW SW 17-28-4 11.44A shows NONE DUE AND PAYABLE AT THIS TIME. Subject to any and all future assessments not yet due and payable.
- 14. Mortgage from Monan Curporation, a Delaware Corporation to American National Bank and Trust Company of Chicago, dated June 23, 1992 and recorded June 25, 1992 at 10:20 A.M. in Mortgage Record 1992, Page(s) 3477-3488 as Instrument No. 92-06-2774 in the Office of the Recorder of White County, Indiana. NOTE: The holder of the Mortgage should be connected for further information.
- 15. Mortgage from Manon Trailer, Inc. to Congress Financial Corporation, dated August 23, 1996 and recorded August 27, 1996 at 11:02 A.M., as Instrument No. 96-08-4052 in the Office of the Recorder of White County, Indiana. NOTE: The holder of the Mortgage should be contacted for further information.

Schedule B: Page 15 of 21

11 9 E 1817 PK

Man 17, 1997 - 3 119W - WUCH SHELIST FOARM

Schedule B Continued:

Commitment No.: 9511-75

- 16. Mortgage from Monan Corporation to Congress Financial Corporation, duted August 23, 1996 and recorded August 27, 1996 at 11:04 A.M., as Instrument No. 96-08-4053 in the Office of the Recorder of White County, Indiana. NOTE: The holder of the Mortgage should be contacted for further information.
- 17. Security Interest of American National Bank and Trust Company of Chicago, secured party, as set forth on a UCC financing statement #5197 signed by T.J. Roshy, president, Manon Corporation, debtor, and Financing Statement recorded March 20, 1987 at 9:30 A.M. and now CONTINUED as financing statement #03-92-481, recorded March 12, 1992 at 10:23 A.M and FURTHER CONTINUED as financing statement 02-97-0289, recorded February 24, 1997 at 10:38 A.M. in the Office of the Recorder of White County, Indiana.
- 18. Security Interest of Amplicon, Inc., secured party, as set forth on a UCC financing statement #11-95-2091 signed by Mike ______, Monon Corporation, debtor, said financing statement recorded November 22, 1995 at 1:53 P.M.
- 19. Security Interest of Congress Financial Corporation (Central), secured party, as set forth on a UCC limineing statement #03-97-0422 signed by _______, IIPA Asset, Inc., debtor, said financing statement recorded March 13, 1997 at 12:32 P.M.
- Intention to Hold Lieu (Mechanie's Lieu) to Monan Corporation, held by R. Michael Muller d/h/a Muller Masonry, dated July 27, 1993 and recorded July 27, 1993 at 2:22 P.M. in Miscellaneous Record 1993 pages 956-960, as Instrument No. 93-07-3173 in the Office of the Recorder of White County, Indiana.
- Intention to Hold Lien (Mechanic's Lien) to Monon Corporation, held by Brehob Corporation, dated January 10, 1996 and recorded January 12, 1996 at 1:14 P.M., as Instrument No. 96-01-0128 in the Office of the Recorder of White County, Indiana.
- 22. Intention to Hold Lien (Mechanic's Lien) to Monon Corporation, held by Mast-Lepley Silo, Inc., dated Murch 4, 1996 and recorded Murch 8, 1996 at 11:40 A.M., as Instrument No. 96-03-1009 in the Office of the Recorder of White County, Indiana.
- 23. Intention to Hold Lien (Mechanic's Lien) to Monon Corporation, held by B&B Air, Inc., duted September 5, 1996 and recorded September 9, 1996 at 9:34 A.M., as Instrument No. 96-09-4254 in the Office of the Recorder of White County, Indiana.
- 24. Intention to Hold Lien (Mechanic's Lien) to Monan Corporation, held by Brim Concrete, Inc., dated September 13, 1996 and recorded September 16, 1996 at 9:09 A.M., as Instrument No. 96-09-4367 in the Office of the Recorder of White County, Indiana.
- Intention to Hold Lieu (Mechanic's Lieu) to Monon Corporation, held by Central Indiana Hardware, Inc., dated October 15, 1996 and recorded October 15, 1996 at 3:05 P.M., as Instrument No. 96-10-4953 in the Office of the Recorder of White County, Indiana.
- Lis Pendens Notice filed November 19, 1993 in the White County Circuit Court, R. Michael Muller d/b/n Muller Musonry versus Monon Corporation, American National Bank and Trust Company of Chicago, and Indiana Oxygen Company, in Lis Pendens Bank 2 page 256.

Schedule B: Page 16 of 21

WALL IN 1997 BIREN WOOR SHELLS FOALK

Schadula B Continued:

Commitment No. 9511-75

- Proceedings pending on a Complaint filed February 7, 1994 in the White County Circuit Court, cause no. 91C(11-9402-CP-012, Computer Graphics Corporation, an Indiana Corporation, plaintiff, versus Monon Corporation, a Delivare Corporation, defendant.
- 28. Proceedings pending on a Complaint filed February 11, 1994 in the White County Circuit Court, cause no. 91CO1-9402-CP-016, Chambers, Galligan & Associates, plaintiff, versus Monon Corporation, defendant.
- 29. Proceedings pending on a Complaint on Account filed March 6, 1995, in the White County Superior Court, cause no. 91DO1-9503-CP-07, All-Place Electric Supply Company, phintiff, versus Monou Corporation, defendant.
- 30. Proceedings pending on a Complaint for damages filed August 11, 1995 in the White County Superior Court, cause no. 91D01-9508-CT-2, Dominic & Sharon Dinicola, plaintiff, versus Monon Corporation, defendant.
- 31. Proceedings pending on a Complaint filed September 16, 1996 in the White County Circuit Court, Cause No. 91001-9609-CP-091, Curter Oil Supply, Inc., plaintiff versus Monon Corporation, defendant.
- 32. Proceedings pending on a Complaint for damages filed October 24, 1996 in the White County Circuit Court, Cause No. 91C01-9610-CP-114, Thomas The tan, plaintiff versus Monan Trailer, Inc., defendant.
- 33. Proceedings pending on a Complaint on Account filed September 16, 1996 in the White County Superior Court, Cruse No. 91D01-9609-CP-35, Brins Cuntrate, Inc., plaintiff versus Monan Corporation, defendant.
- 34. Proceedings pending on a Suit on Account filed September 23, 1996 in the White County Superior Court, Cause No. 91D01-9609-CP-39, Cataphote, Inc., plaintiff versus Monon Corporation, defendant.
- 35. Proceedings pending on a Suit on Account filed September 26, 1996 in the White County Superior Court, Cause No. 91D01-9609-CP-43, Ps iduction Carbide & Steel Company, d/b/a Arknesus Cutting Tools, plaintiff versus Monan Corporation, defendant.
- 36. Proceedings pending on a Complaint filed October 23, 1996 in the White County Superior Court, Cause No. 91D01-9610-CP-48. Midwest Industrial Supply, Inc., plaintiff versus Monon Corporation, defendant.
- 37. Proceedings pending on a Complaint filed October 23, 1996 in the White County Superior Court, Cause No. 91D01-9610-CP-49, Midwest Industrial Supply, Inc., plaintiff versus Monon Corporation, defendant.
- 38. Proceedings pending on a Composint filed July 16, 1996 in the White County Circuit Cause No. 91C01-9607-CP-070, Kenneth J. Zeller, Commissioner Indiana Department of Labor & Michael Hess, plaintiff versus Monon Corporation, defendant.
- 39. Proceedings pending on a Compaint filed July 25, 1996 in the White County Circuit Cause No. 91C01-9607-CP-073, Schneider Specialized Carriers, Inc., plaintiff versus Monon Corporation, defendant.
- 40. Proceedings pending on a Complaint filed July 23, 1996 in the White County Circuit Cause No. 91C01-9607-CP-072, Accuride Caup., plaintiff versus Monon Corporation, defendant.
- 41. Proceedings pending on a Complaint filed May 23, 1996 in the White County Circuit Cause No. 91C01-9605-CP-052, Horion Farms Inc., plaintiff versus Monon Corporation, defendant.

Schedule B: Page 17 of 21

Wat IV 1837 3:12PM MUCH SHELIST FOABLE

Schedule B Continued:

Commitment No.: 9511-75

- 42 Electrical Line Eusement to White County Rural Electric Membership Corporation, dated February 28, 1939 and recorded January 13, 19:0 at 11:30 A.M. in Miscellancous Record "Q" page 48.
- 43 Electrical Lines Eusement to Northern Indiana Public Service Company recorded July 20, 1946 at 10:00 A.M. in Miscellancous Record "U' page 62.
- 44 Electrical Lines Eurement to Northern Indiana Public Service Company recorded July 20, 1946 at 10:10 A.M. in Miscellaneous Record "II" page 64.
- 45 Electrical Lines Easement to Northern Indiana Public Service Company recorded July 20, 1946 at 10:05 A.M. in Miscellaneous Record "U" page 65.
- 46 Electrical Lines Ensement to Northern Indium Public Service Company recorded July 20, 1946 at 10:15 A.M. In Miscellaneous Record "U" page 66.
- 47 Electrical Lines Resembnt to Forthern Indiana Public Service Company recorded January 7, 1957 at 8:30 A.M. in Deed Record 186, page 129.
- 48 Ensurent for Ingress and Egress and for constructing and maintaining a water gate across the Monon Ditch as contained in Warranty Deed dated December 31, 1969 and recorded December 31, 1969 at 2:30 P.M. in Deed Record 223 page 657.
- 49 Ensurent from Motion Corporation to Town of Motion for draining purposes dated July 21, 1986 and recorded August 20, 1986 at 9:30 A.M. in Easument and Right of Way Record 1986 pages 48-49.
- 50. Terms and Conditions of a Memorandum of Understanding and Agreement between Moson Corporation and the Town of Moson, dated July 21, 1986 and recorded August 20, 1986 at 9:35 A.M. in Miscellaneous Record 1986 pages 841-844.
- Easement agreement between Town of Monon and Keith and Annabelle Jackson for construction and maintenance of a 12 inch sewer under portion of Lat 16 on Walnut Street in Bull's Addition dated October 1, 1957 and recorded October 9, 1957 at 9:45 A.M. in Feed Record 190 page 277.
- 52 Easement for Water line granted by Monon Trailer, Inc to Town of Monon dated March 31, 1972 and recorded April 12, 1972 at 2:40 P.M. in Easement and Right of Way Record 1 page 370.
- Easement Agreement between Town of Monon and John Robinson for construction and maintenance of 12 Inch server under portion of Lot 15 on Walnut Street in Ball's Addition dated October 1, 1957 and recorded October 9, 1957 at 9:45 A.M. in Deed Record 196 page 279.
- 54. Electrical Lines Easement from Monon Trailer and Budy Manufacturing Corporation of Indiana to Northern Indiana Public Service Company, an Indiana Corporation, dated January 25, 1969 and recorded February 4, 1960 at 9:00 A.M in Deed Record 201 page 91.
- 55. Ensement Agreement between Town of Monan and Monan Trailer, Inc. dated June 2, 1964 and recorded June 10, 196at 8:00 4.M. in Deed Record 212 page 533.

Schedule B: Page 18 of 21

Marily 1997 Billew Much Sherist Foxeds

Schedule B Continued:

Commitment No.: 9511-75

- 56. Electrical Lines Eastment between Monon Trailer, Inc. and Northern Indiana Public Service Company, an Indiana Corporation, dated June 20, 1963 and recorded August 5, 1963 at 8:15 A.M. in Deed Record 209 page 487.
- 57. Electrical Lines Eusement from Monau Trailer and Body Manufacturing Company to Northern Indiana Public Service Company, an Indiana Corporation, dated December 20, 1956 and recorded February 8, 1957 at 8:30 A.M. in Deed Record 186 page 257.
- 58. Subject to the right of way for County Road 200 West.
- 59. Electrical Lines Eusement from Monon Trailor and Body Manufacturing Company of Indiana to Northern Indiana Public Service Company dated November 6, 1957 and recorded November 23, 1957 at 8:30 A.M. in Deed Record 191 page 173.
- 60. Subject to access to and the use of a well that is located on Lot 60 on Railroad Street as contained in Deed from Monon Railroad to Monon Trailer, lac dated June 29, 1965 and recorded August 9, 1965 at 11:45 A.M. in Deed Record 215 page 467.
- 61. Right of Way for drainings, flow and maintenance of Legal Ditch (or Legal Drain) as set forth in IC 36-9-27-33.
- 62. Subject to a roadway lying East of and adjacent to the C.1. & L. Rullway Right-of-way on the West side. Pertains to .10A in the SW SE 21-28-4.
- 8. NOTE: The acreage indicated in the legal description is salely for the purpose of identifying the said tract and should not be construed as insuring the quantity of land.
- 64. We do not insure access over roads or driveways which are not public.
- 65. Encroachments of the subject real estate and lence encroachments onto admining property, as shown on the Minimum Detail Standard Survey, daten February 24, 1987, prepared by Dennis W. Sterrell, Registered Land Surveyor No. S-0338, and recorded in Miscettaneous Record 1987 page 274 and Plat Book I page 82 on March 11, 1987. In particular, the insured is possessing the following property, but is not the record title helder thereof. Continued use of this property is not insured.

MONON CORPORATION POSSESSES, BUT HAS NO DEED FOR THE FOLLOWING:

- -Lot 49 on Railrand Street in the Original Plat of the Town of Manon lying South of the C.1. & L. Railrand right-
- -Lot 11 on Walnut Street in Benjamin Bull's Addition of the town of Monon lying South of the C.1. & L. Ruilroud right-of-way.
- -Lot 15 on Walnut Street in Benjumin Ball's Addition of the town of Monon lying South of the C.I. & L. Railroad rightof-way.
- -Lat 20 on Spruce Street in Benjamin Bull's Addition of the fown of Manon lying South of the C.I. & L. Railroad right-
- or-way.
 -The land that lies between the South line of Benjamin Bull's Addition of the Town of Monon and the North line of Tract.
 H(a) from Walnut Street to the East line of the Monon Railroad.
- -That part of Walnut Street lying South of the C.I. & L. Roileand right-a -way.
- 66. Subject to the Terms and Conditions of Agreement by and between the Town of Momon and the CSX Railroad Company in the matter of the construction of two new grade crossings over the true's of CSX Transportation, Inc., and removal of a grade crossing at Walnut Street in conjunction with the construction of a new public access road to the Monon

Schedule B: Page 19 of 21

War IT 1887 - Briden - WUCH SHELLST FEARER

Schedula B Continued:

Commitment No.: 9511-75

Corporation, Inc., dated actaber 9, 1989 and recorded June 13, 1990 at 10:32 A.M. in Easement Record 1990 pages 38-55.

- 67. Subject to Ensument from Minion Corporation to the Town of Month, dated February 26, 1990 and recorded June 13, 1990 at 10:32 A.M. in Ensument Record 1990 pages 56-57, as Instrument No. 90-06-2187.
- 68. Environmental Disclosure Document dated April 20, 1990 and recorded April 24, 1990 at 11:07 A.M. in Miscellaneous Record 1990 pages 439-456, instrument no. 90-04-1442 in the Office of the Recorder of White County, Indiana.
- 49. Rights of way for drainage tiles, ditches, feeders and laterals, if any.
- 70. Rights of the Public, the State of Indiana, and County of White and the municipality in and to that part of the premises taken or used for road purposes.
- 71. Liens, if any, on the growing crops on the land.
- 72. Rights of the railroad company servicing the railroad siding located on the insured premises and indicated on survey dated September 6, 1989 prepared by Carl Fuller, Registered Land Surveyor, in and to the ties, rails and other properties constituting said railroad siding or in and to the use thereof.
- 73. Easement for electrical lines dated January 22, 1955 from Amunda Bozich etal to Northern Indiana Public Service Company, its successors and assigns, recorded September 30, 1955 at Deed Record 177, page 233.
- 74. Ensement for electrical lines dated September 2, 1955 from the Lafayette Diocesan Charities, Inc. to Northern Indiana Public Service Company, recorded September 9, 1955 at Deed Record 177 page 73.
- 75. Ensement for Electrical Lines dated February 15, 1957 from John C. Bennett, Bishop, etal to Northern Indiana Public Service Company, recorded August 14, 1957 at Deed Record 189, page 303.
- 76. Right of Way Grant to the State of Indiana for Highway No. 43, dated June 14, 1929 and recorded January 13, 1977 at Eusement Record 1977 page 197.
- 77. Subject to a thirty foot easement as shown in Curporate Warranty Deed from Medica Aggregates Company to Thomas R. Brawn and Patricia A. Brawn, dated April 13, 1979 and recorded May 9, 1979 at Deed Record 1979, page 591.
- 78. Grant of Ensement from Medica Aggregates Company to Evans Transportation Company dated February 6, 1979 granting a non-exclusive permanent ensement, license, right and privilege of passage and use, both pedestrian and vehicular, for purpose of ingress in, to, upon, and over any and all portions of the real estate remaded Murch 2, 1979 at Ensement Record 1979 pages 9-13.
- 79. This policy does not insure any land located between a certain fence and the North property line as depicted an a survey prepared by Carl Fuller, registered land surveyor, dated September 6, 1989. (tract IV)

Schedule B: Page 20 of 21

War 17 1997 3:13PM MUCH SHELLST FEARER

Schedule B Continued:

Commitment No.: 9517-75

- 80. Right of title and interest if any, of owner of property adjoining to the North, and of those chaining through said owner, by adverse possession, because of fence encroachments as shown on narvey prepared by Carl Faller, Registered Lin J Surveyor, dated September 6, 1989. (fract f)
- \$1. Possible lien for taxes on amitted or undervalued property pursuant to IC 6-1,1-9.
- 82. This policy does not lusure legal right of necess to and from the insured premises designated as Tract VI.
- 83. Subject to the reservations and covenants as contained in a special Warranty Deed and Related Covenants dated August 3.9, 1991 by and between Vulcan Materials Company, granter, and Monon Trailer, Inc., granter, recorded August 3.9, 1991 at 3:58 P.M. in Deed Record 1991 pages 1271-1280 as Instrument No. 91-08-3842 in the Office of the Recorder of White County, Indiana.
- 84. Memorandom of Option to Lense by and between Monon Trailer, Inc., a Delaware Corporation, granter, and Vulcin Materials Company, grantee, dated August 29, 1991 and recorded August 30, 1991 at 4:00 P.M. in Miscellancous Record 1991 pages 675-677, as Instrument No. 91-08-3843 in the Office of the Recorder of White County, Indiana.
- 85. Right of Way Grant to State of Indiana for Highway 16, dated April 25, 1935 and recorded December 20, 1976 at 1:. 0 P.M. in Easement and Right of Way Record 1976 Pages 179 & 180.
- 86. Right of Why Grant to State of Indiana for State Road No. 16, dated April 25, 1935 and recorded January 24, 1977 at 8:45 A.M. in Ecsement and Right of Way Record 1977 Pages 460 & 461.
- 27. Terms, provisions and conditions of Memorandom of lease by and between Jan, Inc., an Indiana Corporation, Lessor and Cardinal Communications, Inc., dated Murch 28, 1990 and recorded May 15, 1990 at 11:40 A.M. in Miscellaneous Record 1990 pages 511 thm 514, Instrument No. 90-05-1738 in the Office of the Recorder of White County, Indiana.
- 88. Subject to tenants rights, if may.
- Easement for ingress and egress as contained in Warranty Deed to Monon Corporation dated May 26, 1993 and recorded May 28, 1993 at 3:08 P.M. in Deed Record 1993 page 809, as Instrument No. 93-05-2244. (Parcet D)
- 90. Eusement for gas mains from Floy Templeton to Northern Indiana Public Service Company dated January 18, 1961 and recorded December 22, 1961 at 8:15 A.M. in Deed Record 205 page 489.
- 91. Eusement for Gax Muins from Floy Templeton to Northern Indiana Public Service Company dated Suptember 17, 1962 at 8:00 A.M. in Dued Record 207 page 319.
- 92. Certain matters are specifically excluded from the coverage of this Commitment and Policy. These matters are set for the on "Exhibit A" which is attached hereto and made a part of this Commitment.
- 93. Subject to the terms and conditions of the Subdivision Ordinance of White County, Indiana, as recorded May 19, 19.38 in Miscellaneous Record 1988, Pages 439-485 in the White County Recorder's Office. This Commitment and Policy does not insure compliance with this Ordinance.

- End of Schedule B -

Schedule B: Page 21 of 21

Mar 17 1997 Ballara Mode SHELLST FOARE

Exhibit A

The following matters are expressly excluded from the coverage of this policy and the Company will not pay loss or damage, costs, attorneys' fees to expenses which arise by reason of:

- (a) Any law, ordinance or governmental regulation (including but not kimited to building and zoning laws, ordinances or regulations) restricting, regulating, prohibiting or relating to (i) the occupancy, use or enjoyment of the land; (ii) the character, dimensions or location of any improvement now or hereafter erected on the land; (iii) a separation in ownership or a change in the dimensions or area of the land or any parcel of which the land is or was a part; or (iv) environmental protection, or the effect of any violation of these laws, ordinances or governmental regulations, except to the extent that a notice of the enforcement thereof or a notice of a defect, lien, or encumbrance resulting from a violation or alleged violation affecting the land has been recorded in the public records at Date of Policy.
 - (h) Any governmental police power not excluded by (a) above, except to the extent that a notice of the exercise thereof or a notice of a detect, lien or encumbrance resulting from a violation or alleged violation affecting the land has been recorded in the public records at Date of Policy
- 2 Rights of eminent domain unless notice of the exercise thereof has been recorded in the public records at Date of Policy, but not excluding from

and the last of the contract of the last o

coverage any taking which has occurred prior to Date of Policy which would be binding on the rights of a purchaser for value without Enowledge

- 3. Defects, liens, encumbrances, adverse claims or other matters
 - (a) created, suffered, assumed or agreed to by the insured claimant;
 - (b) not known to the Company, not recorded in the public records at Date of Policy, but known to the insured claimant and not disclosed in writing to the Company by the insured claimant prior to the date the insured claimant became an insured under this policy:
 - (c) resulting in no loss or damage to the insured claimant:
 - (d) attaching or created subsequent to Date of Policy: or
 - (e) resulting in loss or damage which would not have been sustained if the insured claimant had paid value for the estate or interest insured by this policy
- 4. Any claim, which arises out of the transaction vesting in the insured the estate or interest insured by this policy, by reason of the operation of federal bankruptcy, state insolvency, or similar creditors' rights laws.

Exhibit C

NORTHERN DISTRICT OF INDIANA HAMMOND DIVISION AT LAFAYETTE

IN RE:)	
	j	TNIOL NI
	j	ADMINISTRATION
	j	CASE NO. 96-40623
MONON CORPORATION,)	CHAPTER 11
a Delaware corporation,	j	REG/km
TAX ID: 58-1607721).	
)	
MONON TRAILER, INC.)	
a Delaware corporation,)	
TAX ID: 58-2637416)	
)	
Debtors.)	

ORDER APPROVING THE SALE OF SUBSTANTIALLY ALL OF DEBTORS' ASSETS TO HPA ASSET, LLC FREE AND CLEAR OF ALL LIENS, CLAIMS AND ENCUMBRANCES PURSUANT TO 11 U.S.C. §§ 363(b), (f) and (m)

This matter coming to be heard on the Debtors' Motion for Order

Authorizing Sale of Assets (the "Sale Motion") under 11 U.S.C. §363 (b), (f) and (m); notice having been given to all creditors and parties in interest; a hearing (the "Sale Hearing") having been held on the Sale Motion on

______; and the Court having considered all of the pleadings, the statements of Counsel; [and the testimony presented in open Court and the evidence proffered in the record in this case] and good cause appearing;

THE COURT FINDS:

Introduction

1. The findings and conclusions set forth herein and on the record in open court on March _____,1997, constitute the Court's findings of fact and

FEB 20 '97 01:00PM MUCH SHELIST FDF3&R RSSOC.CHICFG

conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding by Bankruptcy Rule 9014. To the extent any finding of fact shall later be determined to be a conclusion of law, it shall be so deemed, and vice versa.

- 2. This Court has jurisdiction to hear this matter pursuant to 28 U.S.C. §§157 and 1334. This is a core proceeding pursuant to 28 U.S.C. §157 (b)(2)(N). Venue is proper in this district pursuant to 28 U.S.C. §§1408 and 1409.
- 3. Proper, timely, adequate and sufficient notice of the Sale Hearing has been given to all creditors, parties in interest and those known to have expressed an interest in purchasing the Debtors' assets.
- 4. All creditors and parties in interest have been provided a reasonable opportunity to object and to be heard regarding the requested relief.

THE SALE PROCESS

- 5. On December 11, 1996, the Debtors sent a bid solicitation to approximately 35 entities and individuals having previously indicated an interest in purchasing the Debtor's assets
- 6. Thereafter, the Debtors received four bids for substantially all of the Debtors' assets.
- After negotiating with three of the potential purchasers, the Debtors determined that the offer submitted by HPA Asset, LLC ("HPA"), reflecting a value of \$14,670,000 (the "HPA Purchase Offer"), was the best offer.

- 8. Subject to Bankruptcy Court approval, the offer made by HPA was formalized by a Purchase Agreement dated February ______, 1997 (the "Purchase Agreement"). The Purchase Agreement is attached as Exhibit "B" to the Sale Motion.
- 9. The Purchase Agreement provides for the sale of substantially all of the Debtors' assets to HPA. The assets (collectively the "Assets") to be sold are enumerated in the Purchase Agreement.
- 10. Pursuant to notice, an Overbid Conference was held in the Office of Thomas Yoder on March ____, 1997. The HPA Purchase Offer was used as the minimum bid and all bidders were required to bid at least \$500,000 more than the HPA Purchase Offer.

Sale Hearing Findings

- 11. Objections to the sale were filed by:

 Such objections were resolved or withdrawn at or prior to the Sale Hearing or otherwise overruled.
- 12. The sale complies with all applicable provisions of the Bankruptcy Code, as required by and §363(b), (f) and (m).
- 13. As required by Bankruptcy Code §363(f), each entity with a security interest in any of the Assets has consented to the sale, is deemed to have consented to such sale or could be compelled in a legal or equitable proceeding to accept a money satisfaction of such interest.

- 14. The Debtors' decision to enter into the Purchase Agreement at this time and sell the Assets is supported by good business purposes and is a sound exercise of business judgment.
- 15. HPA is not an insider of either Debtor, as that term is defined in the Bankruptcy Code.
- 16. HPA is not an affiliate of either Debtor, as that term is defined in the Bankruptcy Code.
- 17. HPA negotiated at arms length and is a purchaser in good faith, as that term is used in 11 U.S.C. §363(m), and has acted in good faith in respect of the Purchase Agreement.
- 18. The HPA Purchase Offer is fair and adequate consideration for the Assets and represents the highest and best value for the Debtors, their estates and their creditors.
- 19. Consummation of the Purchase Agreement is supported by good business reasons and is in the best interests of the Debtors and their creditors.

The Compromise Motion

20. The sale to HPA requires the affirmative support of various parties, including Congress Financial Corporation (Central) ("Congress") (which holds liens on all of the assets proposed to be sold and which is providing financing essential to the transaction), and Rosby Corporation (which is a non-debtor affiliate to the Debtors that holds title to certain intellectual property required by HPA). HPA also conditioned its offer on the affirmative support of the official creditors committee in these cases ("Committee"). In order to obtain the

requisite support of the various parties, Congress, Rosby Corporation, the Committee and the Debtors have entered into a comprehensive Settlement and Compromise Agreement ("Settlement Agreement") that is intended to resolve various issues among those parties relating to the proposed sale. The Debtors have requested that Court's approval of the Settlement Agreement pursuant to a Joint Motion to Compromise Controversies and to Authorize Distribution to Secured Creditors from Sale Proceeds ("Compromise Motion") that the Debtors have requested the Court to consider simultaneously with the Sale Motion. The Court's ruling on the Compromise Motion shall be set forth in a separate order. Because the Committee, Congress, Rosby Corporation and others have relied on the Settlement Agreement in electing not to object to the Sale Motion, this Order shall not be operative unless the Compromise Motion is granted.

IT IS THEREFORE ORDERED, ADJUDICATED AND DECREED THAT:

- A. The Sale Motion is granted in its entirety and the Purchase Agreement and all of the transactions contemplated thereby are approved in all respects.
- B. Any and all objections to the sale were withdrawn or are hereby overruled.
- C. Good and sufficient notice of the Sale Hearing was provided to a creditors and such notice is approved in all respects.
 - D. The Debtors are hereby authorized and directed, pursuant to §363(b),(f) and (m) of the Bankruptcy Code, to sell the Assets to HPA free and clear of any and all liens, claims, encumbrances and interests of any kind or

nature, whether arising prior or subsequent to the filing of the petitions initiating this case, whether imposed by agreement, understanding, law, equity or otherwise, except as provided in the following paragraph for the liens of Congress Financial Corporation (Central), which will continue to encumber the Assets to the extent provided in the Purchase Agreement, with such other liens, claims, encumbrances and interests, if any, attaching only to the proceeds of the sale (the "Proceeds"), with the same validity, priority, force and effect as they now have against the Assets.

- E. In accordance with the terms of the Purchase Agreement, the transfer of the Assets shall be subject to all existing liens and security interests in favor of Congress, which liens and security interests shall continue to secure that portion of the Debtor's indebtedness to Congress that is being assumed by HPA pursuant to the terms of the Purchase Agreement. In addition, Congress shall retain its lien rights against the Debtors and their entities, except that Congress' claims shall be reduced by the amount of Congress' debt that is assumed by HPA.
- F. The Debtors' entry into the Purchase Agreement is hereby authorized and the Debtors are hereby directed to issue, execute, deliver, file and record all documents, instruments and agreements necessary or desirable to consummate the sale to HPA.
 - G. The Purchase Agreement between HPA and the Debtors is approved as being in the best interests of the Estates.

- H. Except as may be otherwise specifically provided in the Purchase Agreement and except as provided herein with respect to Congress, HPA shall not be liable for any claims against the Debtors to the maximum extent provided under applicable law.
- I. This Order shall be binding upon, and inure to the benefit of, the Debtors and HPA, and their respective successors and assigns, including, without limitation, any trustee hereinafter appointed for the Debtors' estates.
- J. Until the cases have been closed, and notwithstanding the closing of the sale transaction, the Bankruptcy Court shall retain jurisdiction over the parties as to the following described matters:
 - a. To enforce all provisions of this Order;
- b. To the extent provided in the Purchase Agreement, to resolve controversies and disputes regarding the interpretation of the Purchase Agreement or any exhibit or schedule thereto; and
- c. To determine such other matters as may be provided for in this Order or as may be authorized under the provisions of the Bankruptcy Code or any other applicable law.
- K. In the absence of a stay pending appeal, if HPA elects to close under the Purchase Agreement at any time on or after the date of entry of this Order, then, with respect to the Purchase Agreement, HPA shall be entitled to the protection of §363(m) of the Bankruptcy Code.

- L. The provisions of this Order shall survive any order dismissing this case or any order or other action converting this case from Chapter 11 to one under Chapter 7 of the Bankruptcy Code.
- M. As allowed by Bankruptcy Rule 8005, notwithstanding Bankruptcy Rule 7062, this Order shall be effective and enforceable immediately upon entry.
- N. The Debtors shall be responsible for preparing and serving the notice of entry of this Order as required by the Bankruptcy Rules; provided, however, that the Debtors need only serve such notice on the parties on the service list.
- O. The provisions of this Order are nonseverable and mutually dependent.

Dated:

United States Bankruptcy Court

Exhibit D

OVERBID PROCEDURES

The procedures for the overbid conference ("Overbid Conference") to sell Debtors' property and assets listed in paragraph 2.1 of the HPA Agreement, entered into by Debtors and HPA on February ______, 1997 ("Debtors' Assets"), shall be on the following terms and conditions, with such terms to apply to all bidders:

- 1. The Overbid Conference will be conducted at the offices of Thomas P. Yoder, Barrett & McNagny, 215 East Berry Street, Fort Wayne, Indiana, on March _____, 1997 at ______p.m.¹
- 2. To qualify to participate in the Overbid Conference, a bidder must deliver a Qualified Bid, as defined below, by 3:00 p.m. C.S.T. on March _____, 1997² to counsel for the Debtors and counsel for Congress at the addresses set forth below.
 - A Qualified Bid is a bid which includes:
 - a. a purchase price of not less than \$15 million plus an agreement to acquire the assets subject to as much as \$170,000.00 of outstanding real property taxes;
 - b. an earnest money deposit of at least \$700,000, which shall be deposited by the Debtors in a separate interest-bearing account created for such purpose subject to an earnest money deposit agreement substantially similar to that entered into by the Debtors and HPA Asset, LLC ("HPA");
 - c. a signed asset purchase agreement (the "Agreement"), subject to no contingencies other than Bankruptcy Court approval, providing for the acquisition or purchase of substantially all of Debtors' Assets. Each Agreement shall contain definitive language providing that if the bidder submitting such Agreement is the successful bidder at the Overbid Conference and thereafter defaults under its Agreement, then that bidder shall forfeit its earnest money deposit;
 - d. a statement disclosing whether the bidder is subject to the Hart Scott Rodino Act (the "Act") and if the bidder believes that it is not subject to the Act, a written statement of legal counsel setting forth the reasons why the bidder is not subject to the Act. If the bid is subject to such filing requirements, the competing bid should clearly state the bidder's intention to pay any filing fees and the bidder's willingness and ability to complete such filing requirements

93618.1:02/20/27

FE3 20 '97 15:23

2194238924

PAGE.19

Afternoon prior to the Sale Hearing.

² At least five days prior to the Sale Hearing.

by no later than three (3) business days after such competing bid is accepted by the Debtors;

- e. a commitment letter from a financial institution or other financing source, or other evidence satisfactory to the Debtors, evidencing the bidder's ability to close within three (3) business days after approval of the Bankruptcy Court; and
- f. the name and facsimile number of the person to whom notice of the purchase prices contained in the various Qualified Bids pursuant to paragraph 4 below should be sent.
- 4. All bidders who have timely delivered a Qualified Bid will be notified by facsimile at least one day before the Overbid Conference of the purchase prices contained in the various Qualified Bids.
- 5. Debtors or Debtors' counsel shall distribute, by same-day or overnight messenger, a copy of the executed HPA Agreement to all parties who have expressed interest in the Overbid Conference and to any party who makes a written request to the Debtors' counsel.
- 6. The Debtors will provide access to their books and records to interested persons for the purpose of conducting due diligence, provided that such persons (a) execute a written confidentiality agreement acceptable to the Debtors, and (b) provide evidence acceptable to the Debtors and Congress Financial Corporation (Central) ("Congress") that such person has the present ability to fund a bid to purchase the Debtors' Assets. All bidders are deemed to acknowledge that they have had an opportunity to inspect the Debtors' Assets and all pertinent documents and related writings prior to making bids, and all bidders are deemed to have relied solely upon their own investigation and inspection in making their offers, except as may otherwise be expressly stated in the Agreement.
- 7. The Assets shall be transferred "As Is" and "Where Is," with no representations or warranties that survive the closing, and no bids shall provide otherwise.
- 8. All bidders shall appear in person at the Overbid Conference, or through a duly authorized representative.
- 9. No other party, other than HPA, will be granted any expense reimbursement, break-up fees, topping fees, or any such type of fee or other bid protection in connection with the Overbid Conference.
- 10. Bidding at the Overbid Conference shall be in \$100,000 cash increments; provided, however, that no bid can be of the same amount as any other bid.

93614.1:02/20/97

- 2 -

- After the bidding has been concluded, the Debtors, after consultation with Congress and a representative of the Unsecured Creditors Committee, will decide whether to accept any bid. If the Debtors accept a bid in accordance with the preceding procedures, the Debtors shall seek Bankruptcy Court approval of such bid at the hearing held upon Debtors' pending Motion to sell substantially all their assets. Within one business day after the Debtors procure such Court approval, the party submitting the approved bid shall increase its earnest money deposit to an amount equal to at least \$1,500,000.
- 12. If the Bankruptcy Court approves a sale but the sale is not timely concluded with the successful bidder, the Debtors may, without any further court order, close on the HPA Agreement or accept any other bid made at the Overbid Conference and consummate the sale with that bidder.
- 13. Congress Financial Corporation (Central) expressly reserves its rights, pursuant to Section 363(k) of the Bankruptcy Code, to credit bid all or a portion of its claim against the assets proposed to be sold.
- 14. All bidders are deemed to have submitted to the exclusive jurisdiction of the Bankruptcy Court with respect to all matters related to the Overbid Conference and the terms and conditions of the transfer of the Debtors' Assets.
- 15. Before the commencement of the Overbid Conference, each bidder or its representatives shall be required to demonstrate to the Debtors and Congress that such bidder or representative is legally empowered, by power of attorney or otherwise, to bid on behalf of the prospective bidder.

Counsel for Debtors

Norman B. Newman
Deborah L. Thorne
Much, Shelist, Freed, Denenberg,
Ament, Bell & Rubcnstein, P.C.
200 North LaSalle Street, Suite 2100
Chicago, IL 60601
(312) 346-3100

Thomas P. Yoder Barrett & McNagny 215 East Berry Street Fort Wayne, IN 46802 (219) 423-9551

93618.1:02/20/97

- 3 -

Counsel for Congress

J. Douglas Bacon Stephen R. Tetro, II Latham & Watkins Sears Tower, Suite 5800 Chicago, IL 60606 (312) 876-7700

Mark Warsco
Karen Courtney
Warsco, Brogan & Strunk
927 South Harrison Street, Suite 200
Fort Wayne, IN 46802
(219) 423-2537

93618.1:02/20/97

-4-

2154238924

PAGE. 22

SCHEDULE G - EXECUTORY CONTRACTS AND UNEXPIRED LEASES

[[Check this box if debtor has no executory contracts or unexpired leases.

DESCRIPTION OF CONTRACT OR LEASE AND NATURE OF
DEBTOR'S INTEREST, STATE WHETHER LEASE IS FOR
NONRESIDENTIAL REAL PROPERTY, STATE CONTRACT NUMBER OF ANY GOVERNMENT CONTRACT.
NUMBER OF ANT GOVERNMENT CONTRACT.
2 Onawa Trucks Lease thru 8/21/97 - \$1,577.12 a month
a comparison of the control of the c
1 7 100 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
1 Forklift - Lease through 5/5/99\$550.00 a month 12 Forklifts - Lease through 10/11/98\$10,597.00 a month
12 Porking - Dease through 10/11/98 \$10,397.00 k month
10 Nissan Forklifts - Lease through 6/13/97\$4,843.62 a month
Lancer Boss - Lease through 10/4/96 - \$1,085.00 a month
Finn Power, Tooling Station, Slug Converyor
Cincinnati Press Brake Lease through 6/13/97 - \$1,408.15 a month
Engineering Modules - Lease through 8/2/99 at \$2,920 a month
Dispensary - Lease through 12/19/95 at \$955 a month Human Resources - Lease through 1/4/96 at \$1,935 a month
Plant 2 & Plant 6 Offices - Lease through 3/28/96 at \$1,066 a month
Finn Power, Tooling Station, PC, Mate Punch Tooling Brake Dies
Lease through 12/31/98 - \$8,949.15 a month
·
Finn Power, Niagara Press Brake Lease through 10/12/98
\$7,918.90 a month
Cincinnati Press, Sheer, Press Brake, Hydraulic Shear Lease through
11/8/98 \$3,408.75 a month Touler Forbild I gare through 12/11/97 at \$2,805.93 a month
Taylor Forklift Lease through 12/11/97 at \$2,805.92 a month
Finn Power Lease through 7/1/98 at \$7,387.73 a month
Boiler, Silo, Gluer, Lexus Lease through 8/25/00 at \$14,450 a month
Sweeper Lease through 12/8/96 at \$757.26 a month
Mita Copier lease through 7/6/00 a \$198.45 a month
Thim copies tease allough Holor & #170.45 & Hollus
· ·

Vėlspar Valspar Painting System Lease through 6/13/97 a \$10,000 a month 1101 Third Street South Minneapolis, MN 554415 USL Capital Welding Robots Lease through 7/01/01 at \$3,452.40 a month 2015 Spring Road, Ste 720 Oak Brook, IL 60521 IBM Leasing Computer Equipment Lease through 6/30/98 \$2,896.60 a month Bank One Center/Tower Computer Equipment Lease through 6/30/98 \$494.00 a month Computer Equipment Lease through 6/30/98 \$253.00 a month 111 Monument Circle, Sye 2100 Computer Equipment Lease through 6/30/98 \$448.00 a month Indianapolis, IN 46204 Computer Equipment Lease through 7/31/98 \$267.00 a month Computer Equipment Lease through 6/30/98 \$1,049.00 a month Computer Equipment Lease through 4/30/98 \$1912.00 a month Computer Equipment Lease through 4/30/98 \$3,127.00 a month Computer Equipment Lease through 5/31/98 \$939.00 a month Computer Equipment Lease through 6/30/98 \$410.00 a month Computer Equipment Lease through 5/13/98 \$5,893.00 a month Computer Equipment Lease through 6/30/98 \$2,118.00 a month Computer Equipment Lease through 2/28/99 \$24,934.00 a month Computer Equipment Lease through 2/28/99 \$1,480.00 a month Computer Equipment Lease through 4/1/99 \$501.00 a month RailRunner Systems, Inc. Master License Agreement (and related Griffin, Georgia agreements)

Three agreements between Monon Corporation and CSX Transportation, Inc.:

- 1. Agreement No. CSX-018673, dated May 1, 1993, covering lease of 15,000 square feet, more or less, of land located at Monon, White County, Indiana, for the purpose of employee parking;
- 2. Agreement No. CSX-024687, dated September 5, 1995, covering a private road crossing located at Monon, White County, Indiana (Station No. 4698+78), for the purpose of vehicular or pedestrian access to Licensee's property.
- 3. Agreement No. CSX-024688, dated September 6, 1995, covering a private road crossing located at Monon, White County, Indiana (Station No. 4722+78) for the purpose of vehicular or pedestrian access to Licensee's property.

SCHEDULE G - EXECUTORY CONTRACTS AND UNEXPIRED LEASES

[[Check this box if debtor has no executory contracts or unexpired leases.

[[Check this box if decitor has h		
NAME AND MAILING AD INCLUDING ZIP CODE, OF OTHER PARTIES TO L CONTRACT.	EASE OR DEB	RIPTION OF CONTRACT OR LEASE AND NATURE OF TOR'S INTEREST, STATE WHETHER LEASE IS FOR LESIDENTIAL REAL PROPERTY, STATE CONTRACT NUMBER OF ANY GOVERNMENT CONTRACT.
	Lease on 17805 Vold Lansing I	NUMBER OF ANY GOVERNMENT CONTRACT. Industrial Building located at: olbrecht Road
·		
	·	

TRADEMARK

REEL: 002091 FRAME: 0728

Schedule 2.2

Cash

Utility deposits of \$25,000

United Parcel deposit of \$13,500

Lease deposits for leases that are not assumed by Buyer

Refunds on Workmen's Compensation programs

Refunds of premiums paid between January 14, 1997 and the Closing Date for automobile, boiler/machinery and general liability insurance policies.

Patents owned by Rosby Corporation

U.S. Patent No. 4,904,017 and pending litigation related thereto (if, but only if, such patent and infringement claims thereunder are transferred by Sellers to Rosby Corporation within thirty (30) days of the Closing Date).

Patent infringement damages on U.S. Patent No. 4,904,017 relating to the period prior to the Closing Date.

Loan receivable in the amount of \$32,000 owing by Railrunner Systems licensor

Intercompany receivable from MTI of Detroit, Inc.

Intercompany receivable from Rosby Corporation

Any refunds of taxes paid by Monon or MTI

Retainers held by lawyers and advisors

Pre-petition recovery claims, including, but limited to:

Small claims actions
Ashland lawsuit
Claim for employee theft
Maxon liftgates
Claim against Henry Price, Esq.
Daubert Chemicals
Highway Metals

Pre-petition receivables of approximately \$10,000 for Indiana withholding taxes

Doof: DS3: 439893.1 2534

Schedule 2.2 (cont'd)

American National Bank & Trust Company of Chicago escrow Rosby Corporation identity not to be used after the Closing Date

Doof: DS3: 439893.1 2534

Schedule 5.2

None.

Mar 7 1997 2:27PM MUCH SHELIST FDABAR

Schedule 5.3

1. Approval of Congress Financial Corporation (Central)

Nº 1788 b 3/18

War. 7. 1997 2:27PM MUCH SHELIST FDABAR

Schedule 5.4 Permits

- 1. Indiana Department of Environmental Management ("IDEM") Construction Permit No. 181-6262-00016
- 2. IDEM Construction Permit No. 181-4989-00016
- 3. IDEM Construction Permit No. 181-4762-00016
- 4. IDEM Construction Permit No. 181-2618-00016
- 5. IDEM Construction Permit No. 181-3664-00010
- 6. IDEM Construction Permit No. 181-3416-00016
- 7. IDEM Operation Permit No. 91-12-91-0091
- 8. IDEM Operation Permit No. 91-12-87-0082
- 9. IDEM Operation Permit No. 91-12-33-0066
- 10. IDEM Registration of one shot blast dated October 19, 1995
- 11. IDEM Registration of one hog (grinder), one surface coating booth and one glue oven dated December 18, 1995.
- 12. NPDES Stormwater permit No. INR 00M 108
- 13. IDEM Office of Air Management Title V Permit

Note: Permit fees and other fees relating to the Business are past due.

#39089v01:Lhb 03/13/97 11:26 AM

Schedule 5.5 Litigation

- 1. See attached "Analysis of Outstanding Legal Issues" dated January 6, 1997 and "Analysis of Pending Legal Issues" dated December 9, 1996
- 2. Indiana Occupational Safety and Health Administration safety orders and proposed fines aggregating \$436,700.
- 3. Commissioner of Labor v. Monon, Docket Nos. 96-75, 96-104 and 96-108, IOSHA Board of Safety Review.
- 4. Kenneth J. Zeller, Commissioner, Indiana Dept. of Labor and Dennis Grogg vs. Monon Corporation. Adversary proceeding in the United States Bankruptcy Court, Northern District of Indiana, Hammond Division at Lafayette, Adversary Proceeding No. 96-4036. Lawsuit alleging discrimination.
- 5. Eugene A. St. Louis and Grace St. Louis vs. Pittsburgh Corning Corporation, et al. Commonwealth of Massachusetts, County of Middlesex, Superior Court, Civil Docket #MIC96-06739S. Personal injury lawsuit for damages.
- 6. Various employment discrimination cases and union grievances.
- 7. Agreed Order OER-92 and Site Participation Agreement relating to the site assessment and remediation of the Four County Landfill Site (Monon is a de minimis PRP with continuing exposure estimated not to exceed \$20,000).
- 8. Indiana Dept. of Environmental Management ("IDEM") Agreed Order H-10358 (relating to the site assessment, remediation and closure plan for Monon's solvent storage pad and solvent storage trailer areas) (April 1991).
- 9. IDEM Agreed Order H-12033 (relating to air violations and hazardous waste violations) (August 1996).
- 10. Alleged Air compliance violations referenced in a letter dated October 4, 1996 from IDEM Office of Air Management.

#35942v01:Lhb 03/13/97 11:32 AM

MONON CORPORATION ANALYSIS OF OUTSTANDING LEGAL ISSUES

MONON - DEFENDANT

Irene Allain Discrimination Alleged discrimination on the basis of an injury to her wrists

Mary McCune Discrimination Alleged discrimination on the basis of a hearing impairment

Domenic Dinicola Party alleges that he suffered a stroke as a result of the pre-employment strength

and agility test. In discovery.

Carpenters Local Union No. 2323 Union alleges that Monon inappropriately discontinued distribution of vending commissions

Four County Landfill Monon has been deemed to have "de minimis" repsonsibility for waste dumped at this site (2,000)

Horton Farms Drainage issue on the land next to the 53 acres

Industrial Building Lease IBL is seeking back taxes from the building leased to MTI.

IOSHA Fines Initial penalty of \$247,000 has been reduced to \$177,000. Still in negotiation with the State

Allen v. Diversified Reclamation Wrongful death suit involving of David Allen.

Dugas v. Monon Personal injury claim resulting from a broken metal bar on rear of truck

Monon tendered this case to its insurance carrier

Martinez v. Gill, et al. Product liability claim involving a chassis.

Zarour v. Monon Wrongful death claim resulting from a traffic accident. Monon and ANR have agreed to settle

ANR's claim for \$10,000. The final paperwork has not yet been executed.

Chambers, Galligan & Associates Chambers is a contingency consultant seeking compensation for personal property tax

services. Chambers lost summary judgment in court. Chambers lost on reconsideration.

U.B. Klem Furniture Property damage claims fo furnitue due an alleged faulty trailer.

Vendor Suits for Past Due Payment See Attachment.

Small Claims See Attachment.

onon Corporation nalysis Of Pending Legal Issues

ctions Filed

		Piaintif Counsel	Monon Coursel		Station.
cuilde Corporation	Accuride is seeking \$336,892.34 for non-payment of outstanding involces for goods sold on open account. Monon shows balance to be \$338,213,74.	Reling Lador & Withored	Much, Shelist, Freed, El Al Ned J. Tonnar	White County Circuit Ct, IN	Active
uminum Line Producis, inc.	Auminum Line is seeking \$372,434.77 for non-payment of outstanding invoices for goods sold on open account. This amount is incorrect per their own list of invoices which total \$319,478.52.	Huffman, Isaac & Hastings	Not Assigned	Cuyahoga County, OH	Active
ıım Concreto	Brim is seeking \$10,159.13 for non-payment of outstanding livokes for goods sold on open account and accrued Interest. Monon shows balance to be \$6,117.05. A Mechanics Llen has been filed, but may not be valid due to the timeliness of filing.	Courtney B. Justice	Ned J. Tonner	White Superior Court, IN	Active.
arter Oil	Carter is seeking payment of outstanding invoices of \$13,615,73 for goods sold on open account. Monon shows a balance to be \$6,743.94.	Dumas & Moriarty	Ned J. Tonnar	White County Circuit	Active
alaphote, Inc.	Calephote is seeking \$25,681.14 for non-payment of outstanding Involces for goods provided on open account.	Rubin & Levin	Not Assigned	White County Superior Ct	Active.
idwest Industrial Supply, Inc.	Midwest has filed two separate actions in White County Superior Court. The first action seeks payment on account of \$5,000 for invoice 25619A. The second action seeks payment of \$17,695.44. Monon agrees with balance due.	Joseph A. Capoliano	Not Assigned	White County Superior Ct	Active
orandal, Inc.	Norandal is seeking \$597.079 for non-payment of outstanding involces for goods sold on open account.	Kirksey & Cron	Much, Sheiir', Freed, Et Al	State Court Of TN	Active
roduction Carbide & Steel b/a Arkansas Cutting Tools	Production Carbide & Steel is seeking \$25,751.45 for goods sold to Menon by Arkansaa Cutting Tools.	Rubin & Levin	Not Assigned	White County Superior	Active
avenswood Aluminum, Corp.	Awarded summary Judgment of \$1,462,034.31 for unpaid Invoices.	Bowles Rice McDavid Graff & Love	Much, Shelist, Freed, Et Al	Circuil Court of Jackson County, WV	Active.
chneider Specialized Carriers	Schneider is seeking damages totaling \$18,330 for wrongfully detaining a trailer. Monon denies detaining the trailer.	Scopelius, Garvin, Light & Hanson	Ned J. Tonner	White Circuit Court, IN	Aciive
equentia	Sequentia is seeking \$57,918,93 for non-payment of outstanding Invoices for goods sold on open account. Monon shows balance of \$43,313,84.	Nicola, Gudbranson & Cooper	Much, Sheli:t, Freed, El Al	Cuyahoga County, OH	Active
loughton Trucking, Inc.	Stoughton is seeking \$23,725 for non-payment of outstanding involces for services provided on open account.	Stoughton In-House Counsel	Much, Sheli:!, Freed, Et Al	Rock County Circuit Court State of Wisconsin	Activ•

dyals Of Pending Hismes

ctions Pending

	\$ 1.00 kg	Pinintif Countel	* Plaintif Countel ** A Monon Countel ** Verye **	Maria and the same and a name	Stature
is Chemical Wasie Mgml		Not Assigned	Not Assigned		Active
ker Specialty & Supply	Baker had three checks returned totaling \$4,992. Baker is threatening to file criminal charges with White County Prosecutor unless payment is received.	Not Assigned	Not Assigned		Active
U Arr, Inc.	B&B is seeking \$90,279.71 for non-payment of outstanding invoices for services and goods sold on open account. Monon shows balance to be \$80,459.04. A Mechanics Llen has been filed, but may not be valid due to the timeliness of filing.	Not Assigned	Ned J. Tonner		Active.
il. Equipment	BGL is seeking return of a \$10,000 deposit placed with Monon Corporation for the purchase of used equipment. Monon has informed BGL that they may use the deposit against remaining equipment.	Arlano, Anderson, Bazos, Hardy & Castillo	Not Assigned		Active
shob Electric	Drehob is seeking \$79,305.85 for non-payment of outstanding involces for services and goods sold on open account, plus interest. They are currently a lienholder and threaten to foreclose.	Not Assigned	Not Assigned		Active
cke Ousiness Forms, Inc.	Cooke Is attempting to collect \$5,345.59 of past due Items. Cooke is utilizing Creditors Collection Bureau	Creditors Collection Buresu	Not Assigned		Active
nnersville Industrial Center	Abandoned Equipment filing for two presses purchased by David Thomas. CIC has a new tenent and wishes the presses removed.	Not Assigned	Not Assigned		Active
awford & Company	Clawford is seeking payment for outstanding service fees. Currently being negotiated by Peterson & Associates, insurance Agent for Monon.	Not Assigned	Not Assigned		Active
in E. Williams Co,	Don E. Williams is seeking damages related to an alleged canceled purchase order. Don E. Williams is unable to produce any signed purchase orders.	Not Assigned	Not Assigned		Active
ial Dynamics	Oual Dynamics is seeking payment of outstanding invoices lotalling \$1,898.77. This amount is not dispuied by Monon Corporation	Milliken & Michaels	Not Assigned		Active.
lon Gorporation	Ealon is seeking \$38,576.00 or return of goods for non-payment of outstanding invoices for goods sold on open account. These goods were shipped prior to 8/22 and the vendor meetings, but arrived 8/23 and Ealon claims they should be treated as COD.	Not Assigned	Not Assigned		Active.

islysts Of Pending Hisues

ctions Pending

	AMERICAN CO.	388	Monon Countel Will Say	/anne-	34701
o i loaing	Eiio Is seeking \$27,885.40 for non-payment of outstanding Involces for goods sold on open account.	Commercial Paralegal, Inc.	Not Assigned Active		Active
sstoria Industries, Inc.	Fostoria is seaking \$5,007.20 for non-payment of outstanding invoices for goods sold on open account.	Dun & Bradstreet RMS	Not Assigned		Active
aygood & Truckpro	Haygood & Truckpro are seeking a balance due of \$2,:29.06.	Milliken & Michaels	Not Assigned		Activ•
erald Journal	Seeking \$121,12 past due for advertising.	Indiana Association of Credit Management	Not Assigned		Active
oosier Rubber & Transmission	Hoosier Is seeking \$6,303.46 for non-payment of outstanding involces for goods sold on open account.	Krieg Devault Alexander & Capehart	Not Assigned		Active
SB Retrability Technologies	HSB Rellability Technologies is seeking payment of work performed prior to the Bankruptcy petition. Has sent Demand letter.	Not Assigned	Not Assigned		Active
slayette Neurology Associates	Monon Corporation stopped payments on two checks paid to this group. One check was for \$329 and the other was for \$323.	Not Assigned	Not Assigned		Active
arshall Steet	Marshall Steel is seeking \$281,424.64 for checks that were not honored by the bank. A second letter was sent concerning an additional check not honored totaling \$12,507.	Not Assigned	Not Assigned		Active
elmor Associales	Meimor is seeking \$5,750.00 for non-payment of outstanding invoices for goods sold on open account.	Dun & Bradstreet RMS Capehart	Not Assigned		Active.
otro-Trader	Metro-Trailer is seeking \$10,992.97 for warranty work performed prior to the Bankruptcy petition. Has sent Demand letter.	Not Assigned	Not Assigned		Active.
ichelin North America, Inc.	Michelin is seeking 1 i, 633,050, i3 for non-payment of outstanding trivolces for goods sold on open account. Threatens collection.	Not Assigned	Not Assigned		Active .
idwest Industrial Supply	Midwest is seeking a payment of \$5,000 for the balance due on the account	Milliken & Michaels	Not Assigned		Active.
ınaı Enterpris a, I nc.	Issued a reclamation notice on October 4, 1996 for goods shipped on Invoice 1975 and 2071. Total amount of the goods in question \$2,970.	Not Assigned	Not Assigned		Active
Islan Of Lafayetta	Oisten is seeking \$7,082.13 for non-payment of outstanding Involces for temporary personnel provided and late charges.	Lauren L. Laderouto Jr.	Not Assigned		Active

ctions Pending

5-19-15-44-08-19	The time of the same of the sa	Manual Court of the Court of th	Monon County	
candi-Wood Floors	Scandt-Wood seeks payment of outstanding Involces for floors which Monon Claims are damaged.	Not Assigned	Not Assigned	Active
harp Steet	Sharp Sleet seeks payment of outstanding Invoices goods shipped on open account or return of said goods.	Not Assigned	Not Assigned	Active
mith Office Plus	Smith Office Plus seeks payment of \$4,268,50 for outstanding service contract invokess on copiers. Has canceled contract due to non-payment.	Not Assigned	Not Assigned	Active.
Lating Kesources, Inc.	Staffing Resources is seeking \$28,483.31 for non-payment of outstanding invoices for services provided on open account and accrued interest.	Lacey, O'Mahoney, Mahoney Sager, King & McCann	Not Assigned	Active
ret Watenduse Co., Inc.	Sited Warehouse seeks payment of outstanding involces goods shipped on open account or return of said goods.	Not Assigned	Not Assigned	Active
wno Air Freight	Towns is seeking payment of outstanding invoices for goods sold on open account.	Alten R. Titlebaum and Associates	Not Assigned	Active
iicy Screen Flocess	Valley Screen is seeking \$7,295.57 for non-payment of outstanding Invoices for goods sold on open account.	Foley & Small	Not Assigned	Active
ncent Metal Goods, Inc.	Vincent is seeking payment of remaining balance of metal they purchased based upon our purchase order,	Hohmann, Boukis & Brunn Co., L.P.A	Not Assigned	Active
eslein American	Western is attempting to collect \$300 on A-335120. Monon paid original invoice of \$355, but Western is now asserting that a revised invoice of \$655 was issued.	Not Assigned	Not Assigned	Active.
Lison international, Lid.	Willson is seeking payment of a freight duty for a shipment from Eile flooting in Canada. They have been directed to seek these moneys from Erie.	Cary N. Staffeld	Not Assigned	Active

Actions Pending

Analysis Of Pendir Jai Issues

Piainiti	Plainiff the second of the part of the second of the secon	Pishilif Counsal	Morion Counts	nuite Statut	tute
				-	
	Pruit is seeking \$9,521.70 for non-payment of outstanding invoices for services provided on open account.	Not Assigned	Noi Assigned	Inaci	Inactive
oul speak					*
year), me.	operdy is seeking 3600.79 selllement of claim.	Interstate Audit Corp.	Not Assigned	Inaci	nactive

Schedule 5.8 Affiliate Assets

- 1. Patent No. 4,904,017 (to be transferred from Monon Corporation to Rosby Corporation)
- 2. See Exhibit "1" attached hereto

#39203v01:Lhb 03/13/97 2:50 PM

ROSBY CORPORATION SCHEDULED PATENTS

February 18, 1997

	U:	SA	Mex	rico	Ca	педа
	File No.	Patent No.	File No.	Patent No.	File No	Patent No.
Interlocking Adapter Casting	3302/16400	4,844,672		Abandoned 5/3/96		Abandoned 5/3/96
Adapter Casting and Interlock	3302/16369	4,925,349	3302/16411	176,501	N/A	N/A
Angle Reinforcement	3302/16401	4,944,421	N/A	N/A	N/A	N/A
Plate Trailer Joints	3302/17344	5,066,066	3302/19516	174,796	3302/19517	2,057,756
Piete Trailer Joints	3302/18844	5,112,099	N/A	N/A	3302/28716	2,141,738
Large Cubic Volume Cargo Container	3302/19728	5,205,428	N/A	N/A	N/A	N/A
Large Cubic Volume Cargo Container	3302/22935	5,248,051	3302/23124	183,261	3302/23123	Serial No. 2,089,865
Thin Gooseneck Assembly	3302/19568	5,322,314	N/A	N/A	N/A	N/A
Variable Height Gooseneck	3302/24198	5,398,956	3302/25983	Serial No. 94 5085	3302/25984	Serial No. 2,123,394
Variable Height Gooseneck (Div. of 3302/24198)	3302/27854	5,456,485	N/A	N/A	N/A	N/A
High-Cube Top Lift Cargo Carrier Structure	3302/24892	5,348,176	3302/26474	Serial No. 94 7410	3302/26472	Serial No. 2,123,394
Intermodal Container with Integrated Interbox Connectors	3302/23597	5,346,084	N/A	N/A	N/A	N/A
Integrated Box Connectors (C-I-P of 3302/23597)	3302/25350	5,462,188	3302/25667	Scrial No. 94 2714		Abandoned 3/4/96

EXHIBIT "1"

TIE TAT-ZORYZZORAJUATACHWIX

21/21 d

Schedule 5.10

None

Nc. 1488 P. 7/16

Mar. 7. 1997 - 2.28PM - MUCH SHELIST FDABAR

Schedule 5.11

Short term catastrophic health insurance

Union pension plan

401(k) Profit Sharing Plan

Agreement between Monon Corporation and United Brotherhood of Carpenters and Joiners (AFL-CIO) Local Union No. 2323 dated July 1, 1994

#39053v01:Lhb 03/13/97 10:22 AM

Schedule 6.4

None

Doc#:DS3:439893.1 2534

SEYFARTH, SHAW, FAIRWEATHER & GERALDSON ATTORNEYS AT LAW

815 CONNECTICUT AVENUE, N.W. - SUITE 500 WASHINGTON, D.C. 20006-4004 (202) 463-2400 FAX (202) 828-5393

ONE CENTURY PLAZA - SUITE 3300 2029 CENTURY PARK EAST LOS ANGELES, CA 90067-3063 (310) 277-7200 FAX (310) 201-5219

1270 AVENUE OF THE AMERICAS - SUITE 2500 NEW YORK, NY 10020-1801 (212) 218-5500 FAX (212) 218-5526

101 CALIFORNIA STREET - SUITE 2900 SAN FRANCISCO, CA 94111-5858 (415) 397-2823 FAX (415) 397-8549 55 East Monrof Street - Suite 4200 Chicago, Illinois 60603-5803

> (312) 346-8000 FAX (312) 269-8869

WRITER'S DIRECT DIAL

269-8895

internet: zvimi@seyfarth.com

February 24, 2000

400 CAPITOL MALL - SUITE 2350 SACRAMENTO, CA 95814-4420 (916) 448-0159 FAX (916) 558-4839

700 LOUISIANA STREET - SUITE 3850 HOUSTON, TX 77002-2731 (713) 225-2300 FAX (713) 225-2340

ONE ATLANTIC CENTER - SUITE 5050 1201 WEST PEACHTREE STREET ATLANTA, GA 30309-3400 (404) 892-0412 FAX (404) 892-7056

INTERNATIONAL

AVENUE LOUISE 500, BOÎTE 8 1050 BRUSSELS, BELGIUM TELEPHONE (32) (2) 647.60.25 FAX (32) (2) 640.70.71

BY EXPRESS MAIL NUMBER: EL514202942US

Comissioner of Patents and Trademarks Box Assignment Washington, D.C. 20231

Re: Recordation of Asset Purchase Agreement

Dear Sir/Madam:

Enclosed are the Recordation Form Cover Sheet - Trademarks Only and the Trademark Asset Purchase Agreement executed between Monon Corporation and HPA Monon, Inc. Also enclosed is our check in the amount of \$90 covering the recordation fee of the same.

In the event of insufficient funds, please withdraw the necessary funds from Deposit Account No. 19-1351.

In addition, we have enclosed a self-addressed stamped postcard. Please place your received stamp on same and return it to our offices. Thank you for your assistance in this matter.

Very truly yours,

SEYFARTH, SHAW, FAIRWEATHER & GERALDSON

By:

MAZ/jhr/1186173

Enclosures

Miles A. Zvi

cc: Bart A. Lazar, Esq.

Certificate of Mailing Under 37 CFR § 1.8(a)

Re: Trademark Asset Purchase Agreement

I hereby certify that the attached correspondence, Recordation Form Cover Sheet (Trademarks Only), Trademark Asset Purchase Agreement and a check in the amount of \$90.00 are being deposited with the United States Postal Service as Express Mail, Express Mail Number EL514202942US, in an envelope addressed to: Commissioner of Patents and Trademarks, Box Assignment, Washington, D.C. 20231, on February 24, 2000.

Janet Hill-Rheams

Person Signing Certificate

Authority

Authority

Person Signing Certificate

Signature

February 24, 2000

Date of Signature



55 East Monroe Street

Suite 4200

Chicago, IL 60603-5803

312-346-8000

fax 312-269-8869

www.seyfarth.com

Writer's direct phone (312) 269-8895

Writer's e-mail mzvi@seyfarth.com

June 15, 2000

BY EXPRESS MAIL Number EL514663167US

U.S. Patent and Trademark Office Assignment Division Box Assignments, CG-4 1213 Jefferson Davis Hwy. Suite 320 Washington, D.C. 20231

Attn.: Kimberly White - Examiner

Assignment Division
Office of Public Records

Re: Recordation of Asset Purchase Agreement by HPA Monon, Inc.

Dear Ms. White:

Pursuant to your letter of May 22, 2000, enclosed please find our resubmitted trademark recordation cover sheet and Asset Purchase agreement between Monon Corporation and HPA Monon, Inc., and a return receipt postcard. Please maintain the original date of filing of the assignment, as we have mailed these documents to you within the 30 day period of your letter and complied with the additional requirements set forth in your letter.

Thank you for your assistance in this matter. Should you have any questions, please call me at the telephone number listed above.

Very truly yours,

Bv

SEYFARTH SHAW

Miles A. Zvi

10132853.1

EXPRESS MAIL CERTIFICATE

Applicant: HPA Monon, Inc.

Re: Trademark Asset Purchase Agreement

Filing Date: February 24, 2000

RECORDED: 02/24/2000

I, Miles A. Zvi, hereby certify that the attached correspondence; Trademark Cover Sheet; Asset Purchase Agreement; and a return receipt postcard, are being deposited with the United States Postal Service by Express Mail, Express Mail Number: EL514663167US, postage prepaid, in an envelope addressed to the U.S. Patent and Trademark Office, Assignment Division, Box Assignments, CG-4, 1213 Jefferson Davis Hwy., Suite 320, Washington, D.C. 20231 on this 15th day of June 2000.

Miles A. Zvi

Person Signing Certificate

Signature

June 15, 2000

Date of Deposit